

Organizational factors, knowledge management and innovation: empirical evidence from medium- and largescale manufacturing firms in Ethiopia

# Citation

ERENA, Obsa Teferi, Mesfin Mala KALKO, and Sara Adugna DEBELE. Organizational factors, knowledge management and innovation: empirical evidence from medium- and large-scale manufacturing firms in Ethiopia. *Journal of Knowledge Management* [online]. Emerald Group Publishing, [cit. 2023-02-06]. ISSN 1367-3270. Available at https://www.emerald.com/insight/content/doi/10.1108/JKM-11-2021-0861/full/html

DOI

https://doi.org/10.1108/JKM-11-2021-0861

Permanent link

https://publikace.k.utb.cz/handle/10563/1011049

This document is the Accepted Manuscipt version of the article that can be shared via institutional repository.



# Organizational factors, knowledge management and innovation: empirical evidence from medium- and large-scale manufacturing firms in Ethiopia

Obsa Teferi Erena, Mesfin Mala Kalko and Sara Adugna Debele

# About the authors

Obsa Teferi Erena is an Assistant Professor at the College of Business and Economics at Hawassa University, Ethiopia. His research interests include corporate governance, earnings management, knowledge management, technology management and financial accounting. He has published in Corporate Governance: The International Journal of Business in Society, Cogent Economics and Finance and African Journal of Science, Technology, Innovation and Development. Obsa Teferi Erena is the corresponding author and can be contacted at: obteferi@gmail.com

Mesfin Mala Kalko is a PhD candidate at the Faculty of Management and Economics at Tomas Bata University in Zlin, Czech Republic. His research interests include corporate governance, corporate social responsibility, corporate finance, innovation management, knowledge management, applied economics and financial accounting. He has published in Corporate Governance: The International Journal of Business in Society, Cogent Economics and Finance and African Journal of Science, Technology, Innovation and Development.

Sara Adugna Debele is a lecturer at the College of Business and Economics at Hawassa University, Ethiopia. Her research interests include financial accounting, corporate governance and corporate social responsibility. She has published in Corporate Governance: The International Journal of Business in Society, Cogent Economics and Finance and African Journal of Science, Technology, Innovation and Development.

## Abstract

**Purpose** – The purpose of this study is to empirically investigate the mediating role of knowledge management (KM) in the linkage between organizational factors, namely, organizational culture (OGCUL) and leadership and management support (LMS) and innovation in medium- and large-scale manufacturing firms in Ethiopia.

**Design/methodology/approach** – A sample of 200 firms has been used to gather data using simple random sampling and to test the proposed hypotheses. Structural equation modeling and crosssectional design were used to analyze the data using LISREL 8.80 SIMPLIS program software tool. Findings - Organizational factors (i.e. OGCUL and LMS) are positively associated with KM and innovation. KM constructs, namely, knowledge sharing, knowledge conversion and knowledge storage, have a significant positive influence on innovation. Knowledge sharing mediates the relationship between organizational factors and innovation.

**Research limitations/implications** – This study has three potential limitations: first, this study is based on a cross-sectional research design. Future research should include longitudinal design to get in-depth insights into the causal inferences. Second, only a few Ethiopian medium- and large-scale manufacturing firms were included in the sample. As a suggestion for future research, other

researchers can include small-scale enterprises using large sample sizes and should examine the effects of organizational factors, KM and innovation across different industries. Finally, this study has only focused on investigating the mediating role of knowledge sharing between organizational factors and innovation. Future research should test the mediating role of the KM process and its constituents (knowledge acquisition, knowledge conversion, knowledge sharing and knowledge storage) between organizational factors and specific aspects of innovation to gain a full understanding of the critical role of KM in organizational innovation.

**Practical implications** – The findings of this study would serve as a guide for policy-makers and managers of manufacturing firms in developing countries in the formulation of policies and long-term strategies. It may also provide a better understanding of the causal relationship between organizational factors, KM and innovation, which in turn has value to directors and managers in manufacturing firms in developing countries as a reference for building a good OGCUL, serving as practical guidance for effective leadership and providing organizational or management support. Specifically, the findings would have the following practical implications: first, firms need to have a combination of KM processes (such as acquisition, storage, sharing and conversion). In practice, developing countries such as Ethiopia have based their innovation strategy on knowledge and technology acquisition through encouraging foreign direct investment. It is not in doubt that Ethiopia has been benefiting from the strategy as a lot of foreign companies have opened their subsidiaries in the country. However, in the authors' view, more emphasis on knowledge acquisition strategy would not take a firm a long time to sustain its innovative activity because it is likely available to firms operating in the same industry, as well as it may hurt a firm's competitive advantage. In addition, by its nature, knowledge may not be retained for future use; it may expire soon. Second, the current highly impulsive and rapid change in the business environment changes the way firms have to operate and deliver products or services. Knowledge (both tacit and explicit) is a resource that can provide a competitive advantage if used well for the intended purpose. In real practice, firms often face challenges in determining where to get knowledge from and how to value or manage it. Besides, knowledge can be obtained from three sources: knowledge can exist in individuals' minds (skills, experience, ideas and insight); knowledge can dwell in a group, which we can call collective knowledge (a team of scientists or researchers); and knowledge can be embodied in an organization's systems, tools, procedures, policies, etc. Knowledge cannot be a valuable resource unless it is obtained and used in designing or producing a product or service. To integrate knowledge with business strategies, there should be a platform or framework that helps to manage it properly. Firm managers, policy-makers and other concerned bodies would consider the three sources of knowledge to foster innovative activities and obtain a competitive advantage. In addition, the authors recommend more emphasis be placed on firm-specific factors (such as OGCUL, leadership, management support and KM) to enhance the innovative capacity of a firm. Finally, the most critical issue to be raised while designing an innovation strategy would be employees' willingness and passion to collaborate with others to develop new ideas, share ideas or implement policies. As knowledge resides in individuals' minds, the knowledge holder should have a passion to share it with those working with him or her. In practice, knowledge sharing depends extremely on the passion and voluntariness of the two parts: knowledge provider and receiver. Therefore, firm managers would design a platform on how to motivate individuals to share their skills, experience and ideas with others through providing incentive packages, punishment and commitment. In this regard, the authors believe that the results would help individuals who are in the position to manage or regulate the manufacturing sector in designing innovation policies, KM policies or technology management policies and business strategies.

**Originality/value** – This study provides new empirical insight into the relationships between organizational factors (such as OGCUL and LMS), KM and innovation in a large sample of firms. To date, the empirical research on these relationships has been mainly limited to descriptive case studies (Chen and Huang, 2009; Zack et al., 2009; Donate and Guadaumillas, 2011), and there is thus a lack of empirical evidence with large samples of firms. Furthermore, there is a scarcity of studies investigating the relationship between organizational factors, KM and innovation in developing countries, especially in Ethiopia. This paper intends to fill this gap and nurture future research studies in the area.

*Keywords:* Organizational culture, Innovation, Structural equation modeling, Leadership, Management support, Knowledge management, The manufacturing sector

Paper type: Research paper

#### Introduction

In this century of the competitive landscape, virtually, all business entities (e.g. startup ventures, major corporations, alliances and others) seek to exploit product market opportunities by using proactive and innovative behaviors (Park et al., 2019). As the business environment is uncertain and dynamic, firms need to learn quickly how to reduce the reverse effect of uncertainty and ambiguity by creating dynamic core competencies to exploit market opportunities. There has been a consensus view in the extant literature (Tidd et al., 2005; Teece, 2000; Zhou and Wu, 2010; Hashi and Stojcic, 2010; Juliao-Rossi et al., 2019) that innovation strategy does have a significant role in sustaining business growth, in particular, and achieving developmental goals, in general. Thus, firms must ensure that their business strategies are innovative to gain and maintain a competitive advantage. Although different authors define innovation in many ways (Schumpeter, 1934; OECD, 2005; du Plessis, 2007), the central concept appears to be the process of developing or exploiting and implementing a new idea or knowledge/behavior that can be transformed into a product/process/method that has a significant market or commercial value (Chin et al., 2014; Neely and Hii, 1998). Besides, any innovation activity seems to add value to a firm in terms of customer satisfaction, market share and competitive edge (Murat Ar and Baki, 2011; Rossi et al., 2019). The term innovation has become an emerging issue in connection to the competitive landscape because it is believed to be a powerful engine that enables a firm to sustain business activity and be competitive (Hashi and Stojcic, 2010). Prior research has confirmed that the degree of innovativeness of industrial firms is a critical factor in their success (Chatzoglou and Chatzoudes, 2018; Cho and Pucik, 2005). However, changing client needs, intense competitive pressure and rapid technological change have made innovation more complex (Tamer Cavusgil et al., 2003). The amount of knowledge available to organizations as a basis for innovation has also increased the complexity of innovation (du Plessis, 2007). As a result, firms are finding it more difficult to internalize innovations (Moorman and Rust, 1999). Nevertheless, innovation is not a oneoff task but rather a gradual process of transforming ideas into marketable products. It would be difficult for the company to be innovative and creative in products and services that allow the company to increase profits and reinvest in the business. On the other hand, a firm's innovative capacity is highly dependent on its ability to integrate, build and reconfigure internal and external competencies (Teece, 2010).

There has been a controversial issue as to what drives innovation in recognizing the role of innovation in fostering a competitive advantage. Prior studies have identified several internal and external factors that enable or obstruct a firm's ability to innovate a product, process or method. Organizational culture (OGCUL), e.g. fosters innovation (Adelekan, 2016; Barney, 1986; Gold et al., 2001; Leonard and Sensiper, 1998). Mokhber et al. (2017), and Carayannis et al. (2003) suggest leadership and

management support (LMS) encourage innovative activity and creative ideas. Moreover, knowledge management (KM) is considered in the literature as a crucial antecedent to innovation (Andries etal., 2019; Khan and Zaman, 2020; du Plessis, 2007). Contrary to popular belief, OGCUL has been identified as the primary impediment to success in organizations (De Long and Fahey, 2000; Rastogi, 2000; Ribiere and Sitar, 2003). Similarly, Ajmal and Koskinen (2008) indicate that OGCUL is a fundamental stumbling block to the success of an organization. Despite the large existing empirical evidence on the drivers of innovation, some other critical issues are not considered yet. For example, organizational factors (such as culture, leadership and management support) might not directly operate well in encouraging innovation but rather through the KM process. The linkage between organizational factors and innovation would be strengthened with the interaction of KM. In this regard, most of the prior studies have focused on examining organizational factors and the KM/innovation relationship or KM and innovation. Although the area of study is traditional with humble empirical evidence, the current study has drawn on how KM mediates the linkage between organizational factors and innovation.

Apart from the extant literature, in 2011, Ethiopia enacted a technology and innovation policy to transfer technology to domestic companies through foreign direct investment. The policy has mainly focused on external barriers to innovation, e.g. shortages of foreign currency, high rates of employee turnover and poor infrastructures such as electricity, roads, water and telecommunication. These issues are common problems for all sectors of the economy, particularly in developing countries (Erena et al., 2021). The policy has ignored the powerful internal antecedents of innovation like organizational factors (such as leadership, management support, culture and structure). Similarly, a five-year (2019-2024) innovation policy of the African Union has emphasized external factors in promoting innovation to sustain industrial growth and considers it a crucial device for building up market share and wealth creation, which may assist companies to survive (Freeman, 1995; Furman et al., 2002). On the other hand, empirical evidence shows that the Ethiopian manufacturing sector has been characterized by low technological knowhow, shortage of skilled manpower, less knowledge about the industry, lower rates of technological products and innovation, weak inter- and intra-sectorial linkages and weak linkages with universities and research institutions (Wakeford et al., 2017; Oqubay, 2018; UNDP, 2018; Erena et al., 2021). UNCTAD (2015) evident that local firms' capacity to take advantage of opportunities both inside Ethiopia and in other emerging markets is hampered by a lack of access to and sharing of research and development (R&D) facilities. To our knowledge, there is no empirical study examining the internal drivers of innovation, such as OGCUL, leadership, management support and KM. This part of an innovative device has been ignored by Ethiopian manufacturing firms. Furthermore, scholars have suggested (Donate and Guadamillas, 2011) the necessity of undertaking subsequent research into the association between organizational factors, KM and innovation to elucidate these relationships and their effects on innovation. In response to their call, we have investigated the relationship between organizational factors (such as OGCUL and LMS), KM and innovation in medium-and largescale manufacturing firms in Ethiopia. Thus, the aim of this study is to empirically investigate the mediating role of KM in the association between organizational factors (i.e. OGCUL and LMS) and innovation using structural equation modeling (SEM) to shed light on the direct and indirect effects of these variables on innovation in medium- and large-scale manufacturing firms in Ethiopia. Specifically, this study seeks to address the following research questions:

RQ1. How do organizational factors such as OGCUL and LMS influence KM and innovation?

RQ2. Does KM mediate the relationship between organizational factors, namely, OGCUL and LMS, and innovation?

RQ2. Can organizational factors such as OGCUL and LMS foster organizational innovation directly?

The remainder of the paper is organized as follows: Section 2 provides an overview of the relevant theoretical framework and the development of hypotheses on the subject. Section 3 describes the research methodology. Section 4 presents the results of the study. Findings are discussed in Section 5. Finally, Section 6 concludes the results and discussions.

# 2. Theoretical framework and hypotheses

The theoretical foundation of this study has been established based on the perspectives of dynamic capabilities (Nonaka, 2007; Teece, 2007; Zhou and Wu, 2010; Tidd et al., 2005; Woodfield and Husted, 2017) and the knowledge-based view (Kogut and Zander, 1992; Grant, 1996; Spender and Grant, 1996). According to the dynamic capability perspective, for sustainable organizational success and competitive advantage, companies should be able to create and combine/reconfigure capabilities in response to changing circumstances (Teece et al., 1997; Teece, 2007), such as innovation capabilities. The perspectives of dynamic KM capabilities (Gold et al., 2001) have been based on two strands of KM capabilities:

- 1. knowledge infrastructure capabilities, consisting of structure, culture and technology as enabling factors; and
- 2. knowledge process capabilities, which include knowledge gathering, conversion, application and protection.

Thus, dynamic KM capabilities perspectives assert that knowledge infrastructure capability improves organizational performance and nourishes knowledge process capability, which enhances organizational innovation (Cho and Korte, 2014). Knowledge, according to the knowledge-based view of the firm, is an organization's most valuable strategic resource. The knowledge-based view of the firm posits that the ability to develop and use knowledge is the most important source of a firm's longterm competitive advantage. It also holds that firms can enhance innovation (Al-Sa'di et al., 2017; Darroch, 2005) and achieve superior performance by effectively and efficiently managing organizational knowledge (Zack et al., 2009). Most knowledge resources are dynamic and intangible by nature, having distinctive qualities that provide sustainable competitive advantage as knowledge offers the foundation for long-term and sustainable differentiation that is difficult to imitate or copy. Despite their complexity, Martelo-Landroguez and Cegarra-Navarro (2014) argue that the integration and alignment of intangible resources like knowledge resources in a firm are critical to innovation. Innovation has been viewed as a critical element for dealing with the problems of uncertainty and stiff competition, as well as gaining a competitive advantage to assure survival and success in the global market (Vargas, 2015). Failure to innovate can jeopardize a company's long-term viability. As a result, firms place a high priority on cultivating an environment that encourages employees to innovate (Shanker etal., 2017). Prior research identified several key antecedent factors that influence innovation, such as OGCUL (Murat Ar and Baki, 2011; Lemon and Sahota, 2004; Valencia etal., 2010; Do et al., 2018; Pedersen etal., 2018; Tellis, 2012), leadership (Makri and Scandura, 2010; Murat Ar and Baki, 2011; Rosing etal., 2011), top management support (Murat Ar and Baki, 2011), KM (Darroch and McNaughton, 2002; Khan and Zaman, 2020; Liao and Wu, 2010) and knowledge sharing (KSHR) (Lei et al., 2020; Ritala et al., 2018; Singh etal., 2019), that enhance and facilitate innovation.

Thus, by integrating the perspective of dynamic capabilities and the knowledge-based view, we argue that organizational factors such as OGCUL and LMS can positively influence KM processes (such as knowledge acquisition (KAC), knowledge conversion (KCOV), KSHR and knowledge storage (KSTR)), which in turn improves organizational innovation in medium- and large-scale manufacturing firms in Ethiopia. In the context of this study, organizational factors (such as OGCUL and LMS) and KM were considered antecedent factors of innovation. Moreover, this study also explores whether organizational factors (i.e. OGCUL and LMS) affect KM, with an emphasis on the mediating role of KSHR in the relationship between organizational factors and innovation. The proposed conceptual model for the study is portrayed in Figure 1.

# 2.1 Organizational culture and knowledge management

OGCUL is defined as the shared values, beliefs (Sackmann, 1992) and practices (De Long and Fahey, 2000) of the people in the organization. It is believed to be the most significant input to effective KM (Alavi and Leidner, 2001; Gold et al., 2001; Leonard, 1995). It encourages dialogue between individuals or groups or among employees, which is often the basis for the creation of new ideas and, thus, creating knowledge (Gold et al., 2001). This interaction among employees helps to transform tacit knowledge into explicit knowledge, thereby transforming it from the individual to the organization level (Nonaka 1990, 1994; Nonaka, and Konno, 1998). People would share ideas and insights in an organization with a knowledge-sharing culture because it is natural for them, rather than something they are obliged to do (McDermott and O'Dell, 2001).



Figure 1 Conceptual model

In a similar vein, Guaderrama (2016) indicated that the lack of a sharing culture impedes KSHR and transfer in organizations.

The empirical literature on the relationship between OGCUL and KM yielded inconclusive results. One stream of literature finds OGCUL is positively associated with KM, whereas other studies indicate no such linkage. Because it influences how members acquire, learn and share knowledge, OGCUL is

regarded as a critical factor in establishing and reinforcing knowledge creation and KM in organizations (Alavi and Leidner, 2001; Gupta and Govindarajan, 2000). Zheng etal. (2010) found OGCUL has a strong positive influence on KM. The authors further suggested that KM practices need to center on incorporating culture-building activities to foster an environment that is knowledge-friendly. Contrary to popular belief, OGCUL has been identified as the primary impediment to successful KM in organizations (De Long and Fahey, 2000; Rastogi, 2000; Ribiere and Sitar, 2003). Similarly, Ajmal and Koskinen (2008) indicate that OGCUL is a fundamental stumbling block to the success of the KM process. Thus, we posit that:

H1. OGCUL is positively associated with KM.

# 2.2 Organizational culture and innovation

The influence of OGCUL on innovation has been recognized from the perspective of the dynamic knowledge capabilities underpinning it (Nonaka, 2007; Tidd etal., 2005; Woodfield and Husted, 2017). The theory of dynamic knowledge capability stipulates that efficient operations result from the sharing of both tacit and explicit knowledge, which in turn contributes to a shared/collective understanding of how things function and could work. This is similar to how culture is defined as how we do things around here. Nonaka (2007) and Woodfield and Husted (2017) also indicate that individual employees at all levels of the organization rely on personal tacit knowledge, as well as subjective insights, intuitions and hunches/clues. Thus, only by establishing an appropriate OGCUL that supports innovation will these be made available for testing and usage by the entire organization.

The relationship between OGCUL and innovation has been extensively studied in the empirical literature, but the results are indecisive. Some scholars indicate OGCUL encourages innovation/creativity (Adelekan, 2016; Barney, 1986; Gold et al., 2001; Leonard and Sensiper, 1998), while others suggest it either promotes or impedes innovation (Li et al., 2018; Naranjo-Valencia et al., 2011; Valencia et al., 2010). OGCUL can be a source of competitive advantage if it is valuable, rare and difficult for competitors to copy or imitate (Barney, 1986), a key success factor in the KM process (Davenport et al., 1998) and a catalyst for continuous innovation (Gold et al., 2001). In a similar vein, Leonard and Sensiper (1998) pointed out that OGCUL would play a significant role in encouraging interaction between individuals, which is essential in the innovation process. Wan et al. (2005) also indicate that for employees to be driven to innovate, there must be a culture that encourages them to do so. Skerlavaj et al. (2010) also found OGCUL has a positive impact on innovation. These positive impacts appear especially strong for technical and administrative innovations (ADMINs). Hassan et al. (2012) also found a positive relationship between OGCUL and innovation. By contrast, Valencia etal. (2010) suggest OGCUL is thought to be one of the most important factors in both promoting and limiting creativity/innovation. Similarly, Naranjo-Valencia et al. (2011) indicate OGCUL can encourage or discourage innovation, which may affect the overall performance of the firm. If an organization's culture is rigid, closed and difficult to conform to, it may resist new ideas, stifling creative minds and inhibiting innovation (Li et al., 2018). Furthermore, Lemon and Sahota (2004) indicate innovation cannot be considered separately from the culture that facilitates or constrains the ability to add value because it is holistic in nature.

Based on the empirical results, we posit that:

H2. OGCUL is positively related to innovation.

# 2.3 Leadership and management support and knowledge management

Langton et al. (2013) defined leadership as the ability to influence a group toward the accomplishment of a vision or set of goals. Leadership is crucial in KM efforts. Leadership is one of the most important factors in the management of knowledge assets and the growth of knowledge and success of knowledge processes in a firm. Crawford (2005) found a strong link between leadership (transformational) and KM because leaders encourage information acquisition, creation and application. Similarly, top management ensures effective KM and resource allocation to boost productivity and innovation processes (Rajan et al., 2021). Top management also promotes organizational effectiveness, KM and team learning to stimulate innovation within the firms (Masa'deh et al., 2016). Management support is also viewed as a critical factor in KSHR (Connelly and Kelloway, 2003) and is fundamental in creating a culture conducive to knowledge creation and dissemination (Lin, 2006). Lin (2007) also indicates that top management support is necessary for knowledgesharing practices. Similarly, Ye§il etal. (2013) pointed out that to create a supportive climate and provide sufficient resources for organizational KSHR, top management support is essential. Drawing on these empirical findings, we propose:

H3. LMS is positively associated with KM.

# 2.4 Leadership and management support and innovation

Leadership is crucial in innovation activities. Leadership plays a critical role in supporting innovation by influencing firm strategic decisions, policies and procedures, and it is a key agent for promoting changes/creativity in the firm that supports innovation (Prasad and Junni, 2016). Similarly, leadership is directly linked to innovation because of its wide-ranging effects on the entire firm's operations, and it is also indirectly linked to innovation because it fosters a positive knowledge-sharing climate that encourages creativity and innovation (Lei et al., 2020). Senge (1990) also demonstrated a positive view toward innovation among leaders is necessary for implementing and developing innovation within the firm. De Jong and Den Hartog (2007) indicated that employees' innovative behavior is influenced by leaders' deliberate actions aimed at stimulating idea generation. The theoretical support for the influence of LMS is based on the perspective of the dynamic capabilities of firms, which states that the ability of an organization to innovate and learn will be influenced by the management styles used and employed (Teece, 1996, 1998).

Prior studies have confirmed a positive relationship between leadership and innovation. For example, Mokhber et al. (2017) suggest that transformational leaders encourage innovative activity within the company and also ensure that the innovations are commercially successful. Redmond etal. (1993) demonstrated that leaders (transformational) may build circumstances that inspire followers to innovate by setting group goals and managing essential resources. Singh etal. (2019) also revealed a positive effect of shared leadership and a market-oriented culture on firm innovation capability. Furthermore, Carayannis et al. (2003) revealed that management support, which is willing to take risks and encourage creative ideas, is one of the catalysts of innovation. Drawing on these empirical findings, we propose:

H4. LMS is positively associated with innovation.

# 2.5 Knowledge management and innovation

The link between KM and innovation has received much of the researchers' attention. Knowledge is widely recognized as an important weapon for sustaining competitive advantage (Lee and Choi, 2003). KM is defined as the process of acquiring, creating, sharing, storing and using knowledge (Nonaka et al., 2000). Gloet and Terziovski (2004) also describe KM as the formalization of and access to experience, knowledge and expertise that build new capabilities, enable higher performance, foster innovation and increase customer value. It is considered a crucial antecedent to innovation (Andries et al., 2019; Khan and Zaman, 2020). du Plessis (2007) indicates that KM fosters a knowledge-driven culture in which to incubate innovations. The author further states that the sharing of knowledge is enhanced by a culture that values knowledge, KM, innovation and creative thinking. Theoretical support for KM is promoted from the perspective of dynamic knowledge creation theory, which states that a shared knowledge base increases knowledge creation within society (Nonaka et al., 2000). A decent number of prior studies on KM indicate that KM is positively related to innovation (Noruzy et al., 2013; Jimenez-Jimenez et al., 2014; Pawlowsky and Schmid, 2012; Liao and Wu, 2010; Darroch, 2005; Nonaka and Takeuchi, 1995; Lee and Choi, 2003; Tamer Cavusgil etal., 2003; Chang and Lee, 2008; Andreeva and Kianto, 2011). For example, Darroch (2005) and Liao and Wu (2010) indicate that KM is positively linked to organizational innovation. Using an open innovation lens and taking a sample of 129 firms operating in a wide array of sectors in Italy, Papa et al. (2020) found a positive relationship between KAC and innovation performance. Similarly, Noruzy etal. (2013) revealed that KM directly influences organizational innovation. Hence, this study puts forward the following hypothesis:

H5. KM is positively related to innovation.

# 2.6 Organizational culture, knowledge sharing and innovation

KSHR refers to making relevant knowledge available to co-workers in the firm (Grant, 2016; Lin and Lo, 2015; Wang et al., 2014; Zhang and Jiang, 2015) to achieve innovation at the individual level (Bavik et al., 2018; Huang et al., 2014) and organizational level (Donnelly, 2018; Oyemomi et al., 2018). Alhady et al. (2011) indicate an organization that encourages employees to contribute knowledge (within groups and at firm levels) is more likely to create new and better ideas, as well as new business prospects and opportunities, hence facilitating organizational innovation. Similarly, prior studies (Donate and Guadamillas, 2011; Gold etal., 2001) indicate an OGCUL that has values oriented toward openness, collaboration and trust is more likely to foster employees' sharing of more ideas and knowledge, which means they are more likely to be innovative, responding more effectively and swiftly to changes and new market opportunities. The dynamic capability theory (Nonaka and Takeuchi, 1995) also claims that innovation occurs when employees share their knowledge, and shared knowledge leads to new or novel insights about creativity and innovation (Sheng, 2017). Previous research (Chang et al., 2017; Yang et al., 2018) suggests that KSHR mediates the relationship between OGCUL and innovation. For example, Chang et al. (2017) found KSHR mediates the association between OGCUL and innovation capability in the knowledgeintensive automobile industry in Taiwan. Furthermore, their findings revealed that OGCUL has a significant positive effect on KSHR. In a similar vein, using data collected from 77 Chinese firms, Yang etal. (2018) find that KSHR plays a mediating role between culture (collaborative) and innovation, especially with respect to product innovation (PDI) and process innovation (PCI). Hu etal. (2009) also indicated that organizations must first create/develop knowledgesharing behaviors as well as a better team culture to attain/achieve high-service innovation performance. Based on empirical studies, we propose that:

H6. KSHR mediates the relationship between OGCUL and innovation.

# 2.7 Leadership and management support, knowledge sharing and innovation

According to the dynamic capabilities perspective, leadership is a catalyst for the development of dynamic capabilities (Nonaka etal., 2016) and is one of the most important contextual resources that foster firms' innovation (Sheehan etal., 2020). Some studies have also supported the perspective that top management has a significant role in a company's dynamic capabilities (Helfat and Peteraf, 2015; Helfat etal., 2007). Successful knowledgesharing practices allow businesses to increase their knowledge capital and exploit and convert all available resources into dynamic competencies like innovation (Le and Lei, 2019; Darroch, 2005). Prior studies (Masa'deh etal., 2016; Zheng etal., 2017) indicate KSHR mediates the relationship between leadership and key organizational outcomes such as innovation. Using data collected from 394 participants at 88 Chinese firms, Le and Lei (2019) found leadership (transformational) positively impacted KSHR as well as PDI and PCI. Furthermore, they indicate that KSHR mediates leadership (transformational) effects on innovation. In a similar vein, Bass and Riggio (2012) demonstrated that leadership (transformational) can influence KSHR and innovation by causing followers to behave in a goal-oriented manner, hence enhancing organizational innovation and performance. Several prior studies (Choi etal., 2016; Xiao etal., 2017; Le etal., 2018) also indicate that individuals share their key knowledge, which is the source and basic driver of enhancing a firm's innovation capabilities, facilitated through leadership (transformational). Furthermore, Choi et al. (2016) found that a firm's ability to acquire and use knowledge mediates the association between leadership (transformational) and innovation behavior. Noruzy et al. (2013) also found that transformational leadership directly and positively influenced manufacturing firms' KM and organizational innovation. The authors further indicated that leadership (transformational) positively and indirectly influenced innovation through KM. Based on the above discussion, we hypothesize that:

H7. KSHR mediates the relationship between LMS and innovation.

# 2.8 Conceptual model

This study has developed seven hypotheses based on the theory of dynamic KM capabilities, the knowledge-based view and the empirical literature on innovation, and now proposes a framework for the relationships between the concepts discussed (Figure 1). Innovation is the dependent variable; KM is the mediator; the independent variables are OGCUL and LMS.

# 3. Methodology

# 3.1 Sample and sampling techniques

With regard to the target firm category, we followed the Ethiopian Central Statistical Agency (2014) benchmarks to classify firms into medium- and large-scale manufacturing firms. Accordingly, firms with more than ten but less than 51 employees are classified as medium scale, whereas firms with 51 or more employees are classified as large scale (Kalko et al., 2022). This study focuses on medium- and large-scale manufacturing firms because medium and large-scale firms and firms in the manufacturing sector are more likely to engage in innovative activities (Daksa et al., 2018). We obtained data on a list of medium-and large-scale manufacturing firms from the Ministry of Industry and the Central Statistical Agency. The data showed that up to the end of the year 2018, 3,520 medium- and large-scale manufacturing firms were registered and operating in Ethiopia (Erena etal., 2022). A total of 200

firms were chosen at random for this study to collect the relevant data. The simple random sampling technique has been used because it gives an equal chance to the firms under study, which would in turn reduce sample bias. There is no consensus in the literature on how large the sample size should be to get a good model fit with SEM. Several Monte Carlo studies (Anderson and Gerbing, 1984, 1988; lacobucci, 2010) havedemonstrated that SEM can perform well with sample sizes of 100 or more. Hair etal. (2019) proposed that the number of factors would be five or less with more than three indicators per factor for a sample size of 100 to 150. With non-under-identified constructs, the maximum number of factors for a sample size of 150 to 300 would be seven. Furthermore, Wolf et al. (2013) suggest that the number of factors, the number of indicators and the strength of the indicator loadings affect the sample size. This means that as the number of factors in the model increases, so does the minimum sample size required.

# 3.2 Measures

Our multi-item scales were drawn generally from prior published studies. The OGCUL and LMS measuring instruments consist of eight items adopted from Kulkarni et al. (2007) and Mageswari (2014). Those items were operationalized as a five-point Likert scale ranging from "1" (strongly disagree) to "5" (strongly agree). KM was measured by using the items developed by Mageswari etal. (2017) and Gold etal. (2001). The KM dimensions scale had 14 items consisting of four interrelated processes such as KAC, KCOV, KSHR and KSTR. A five-point Likert scale ranging from "1" (very low) to "5" (very high) was used. The measures of innovation dimensions (product, process, administrative and marketing) were adopted from Hsu et al. (2014), Prajogo and McDermott (2011), Prajogo and Sohal (2006) and Vicente et al. (2021). This scale had 17 items measured using a five-point Likert scale ranging from "1" (nothing was done) to "5" (very good) that reflects firm quickness to generate novel or new ideas, new product launching, new product development, new processes, new technologies and new methods of problem-solving against key competitors. Before final administration, questionnaires were used in a pilot study. 25 manufacturing firms completed the questionnaire and provided helpful comments to examine the reliability, validity and usefulness of the measurement instruments in terms of language clarity, coherence and appropriateness. Based on the respondents' feedback, we prepared changes to the questionnaire to improve its precision and suitability. Professional data enumerators were hired and trained. Finally, 200 physical copies of the survey questionnaires were handed out face-to-face to reduce the number of people who declined to participate and enhance the response rate. The firm's middle manager completed and returned 153 of the 200 surveys that were provided. The questionnaire was distributed and collected during the year 2018. Overall, the response rate was 76.50%.

# 3.3 Data management and analysis

The nature of the study is quantitative, and it used cross-sectional observations, which assume observation over a single period across various firms. Co-variance-based SEM was used to assess the fit of the hypothesized model using the LISREL 8.80 program. We assessed the validity and reliability of the measurement model using confirmatory factor analysis. Traditionally, reliability is described as the correlation between the indicators of a latent variable (Bollen and Lennox, 1991). A model's reliability involves both composite reliability (CR) and individual indicator reliability. The CR assesses internal consistency, which determines whether the items measuring a latent are similar in their scores. The recommended level of acceptable fit is above 0.70 (Bagozzi et al., 1991). The individual indicator reliability is standardized loading or outer loading; it indicates how much of the variation of

the item is defined by its latent construct. The rule of thumb threshold for the test is above 0.70. However, high reliability does not mean that the model is appropriate, so validity must be assessed.

Validity was tested using two methods: convergent validity and discriminant validity. The average variance extracted or retrieved (AVE) is used to determine convergent validity. Discriminant validity was tested using the correlation coefficients between latent variables. If the square root of each latent variable's AVE surpasses the correlation coefficient between the latent variables, discriminant validity is satisfied (Fornell and Larcker, 1981). The goodness of fit test is another key topic in SEM. The traditional statistical tool or method for testing absolute fitness is x2, with the null hypothesis that there is no statistically significant difference between the sample covariance matrix and the covariance matrix produced from a theoretically specified model or population covariance. The presence of a nonsignificant x2 suggests that the two co-variances are similar and that the model fits the data. The model does not fit the data if the x2 value is significant (a = 0.05). Steiger and Lind (1980) created root mean square error of appropriation (RMSEA) as a complement to x2 RMSEA is a measure of the average standardized residual per degree of freedom. Steiger and Lind (1980) proposed that RMSEA values below 0.05 indicate a very good model fit (close fit), and values less or equal to 0.10 suggest a reasonable or satisfactory fit, but a value greater than 0.10 shows some misfit between the model and the data. However, empirical research (Bollen, 1987; Browne and Cudeck, 1989; Cheung and Rensvold, 2002; Fan et al., 1999; Fornell and Larcker, 1981; Loehlin, 2004; Schumacker and Lomax, 2016) shows that the x2 test has various flaws, including sample size sensitivity and the lack of a stated power function. As a result, researchers devised other goodness of fit indicators in response to the chi-square test's limitations, such as goodness-of-fit index (GFI), adjusted GFI (AGFI), comparative fit index (CFI), normed fit index (NFI), non-normed fit index (NFI), and normed (x2/df). In general, a level of acceptable fit of 0.90 or greater is recommended for GFI, CFI, NFI and NNFI.

# 4. Results

## 4.1 Measurement models

The main objective of this study is to examine the mediating role of KM in the relationship between organizational factors and innovation. Three measurement models (organizational factors, KM and innovation) were separately assessed for validity and reliability using confirmatory factor analysis in LISREL 8.80, a SIMPLIS program software tool. In this study, OGCUL and LMS were used to represent organizational factors. KM constructs involve KAC, KCOV, KSHR and KSTR. The study used four constructs of innovation, namely, PDI, PCI, administrative innovation (ADMIN) and marketing innovation (MRIN). The results, summarized in Table 1 and Figures A1-A3 in Appendix 1, showed the CR values for all constructs exceeded the minimum standard norms of 0.70, as recommended by Fornell and Larcker (1981) and Hair et al. (2019). Specifically, the lowest CR value of 0.791 (for KCOV) was from the KM measurement model, whereas the highest CR value of 0.878 (for PCI) appeared in the innovation measurement model. These results indicate that all scales are inhabited with adequate internal consistency reliability.

In this study, we used item loading of more than 0.6 and significant at 0.01 to ensure convergent validity. As a result, each item loading presented in Table 1 is greater than 0.60 and significant at 0.01, which suggests convergent validity is satisfied. In addition, the AVE was manually computed to test convergent validity, and the result in Table 1 showed that each latent variable had been observed with an AVE greater than the minimum acceptable level of 0.50, recommended by Bagozzi and Yi (1988). This result also supports the convergent validity of the constructs in all models. Discriminant validity can be evaluated using two methods based on the criteria set by Fornell and Larcker (1981). The first

method is by computing the correlation coefficient between two constructs and comparing the result to the AVE of each construct. If the AVE of each construct is greater than the computed correlation coefficient, discriminant validity is satisfied. Otherwise, the constructs should be merged to yield a single factor. The second approach is to compute the square root of the AVE for each construct and compare it with the shared correlation coefficient between the constructs. The value of the square root of the AVE for each construct should exceed the correlation coefficient to support discriminant validity. Table 2 presents the correlation of construct variables and the square root of AVE on the diagonal. The result shows that discriminant validity is satisfied for all construct variables.

Construct	ltem	Mean	Std. dev	Standardized	CB	AVE
Construct		moun		iouonig (n)	011	7.02
OGCUL	OGCUL1	2.738	0.998	0.62	0.864	0.564
	OGCUL2	2.882	0.938	0.71		
	OGCUL3	2.980	0.899	0.90		
	OGCUL4	3.091	0.913	0.76		
	OGCUL5	3.085	0.972	0.74		
LMS	LMS1	3.098	1.080	0.61	0.830	0.555
	LMS2	3.000	1.025	0.76		
	LMS3	2.777	1.040	0.91		
	LMS4	2.921	0.949	0.67		
KAC	KAC1	2.882	0.924	0.78	0.863	0.617
	KAC2	2.882	0.993	0.90		
	KAC3	3.078	0.899	0.82		
	KAC4	3.156	0.987	0.61		
KCOV	KCOV1	2.705	0.902	0.75	0.791	0.560
	KCOV2	2.817	0.983	0.80		
	KCOV3	2.888	1.003	0.69		
KSHR	KSHR1	3.124	0.961	0.67	0.837	0.565
	KSHR2	3.091	0.941	0.77		
	KSHR3	2.947	0.985	0.76		
	KSHR4	3.032	0.941	0.80		
KSTR	KST1	2.862	0.932	0.74	0.817	0.600
	KST2	2.856	0.989	0.85		
	KST3	2.902	0.951	0.73		
PDI	PRDIN1	2.790	0.863	0.67	0.849	0.590
	PRDIN2	2.823	0.980	0.93		
	PRDIN3	2.875	0.975	0.72		
	PRDIN4	2.888	0.942	0.72		
PCI	PRCIN1	2.875	0.988	0.80	0.878	0.650
	PRCIN2	2.882	1.081	0.87		
	PRCIN3	2.934	1.086	0.90		
	PRCIN4	3.006	0.983	0.61		
ADMIN	ADMIN1	2.790	1.030	0.67	0.868	0.625
	ADMIN2	2,830	1.062	0.93		
	ADMIN3	2.954	1.065	0.79		
	ADNIN4	3.013	1.019	0.75		
MRIN	MRIN1	3.006	1 103	0.72	0.857	0.546
	MRIN2	2 973	1.038	0.72	0.001	0.040
	MRIN3	3.032	1,102	0.76		
	MRIN4	3.039	1 044	0.77		
	MRIN5	3.058	1.046	0.67		
	WI III YO	0.000	1.040	0.07		

#### Table 1 Summary statistics of the measurement model

Notes: Composite reliability (CR) =  $\frac{\left(\sum_{i=1}^{n} ti\right)^{2}}{\left(\sum_{i=1}^{n} ti\right)^{2} + \sum_{i=1}^{n} t^{me}}$  Where  $(fl)^{2}$  - the squared sum of standardized loadings.

 $\sum_{i=1}^{n} me - \text{the sum of measurement error (unexplained variance of an indicator by its latent variable).}$ AVE is manually computed as the sum of the variances of indicators explained by a latent construct divided by the number of indicators. Explained variance is squared standardized loading (correlation) or 1 – measurement error (unexplained variances). It is expressed as  $AVE = \frac{Sum of explained variances}{Sum of explained variances}$ 

The CR of the second order shown in Figure A4 in Appendix 1 was higher than the minimum standard of 0.7, indicating adequate internal consistency reliability. The fit index statistics obtained acceptable results, which suggests the conceptual model best fits the data. KM (second-order construct) has a strong significant positive relationship with all constructs (first-order constructs): KAC, KCOV, KSHR and KSTR. The results indicate that the four constructs are important indicators of KM and support the first-order model or measurement model.

No.	Variable	1	2	3	4	5	6	7	8	9	10
1	OGCUL	0.75									
2	LMS	0.55	0.74								
3	KAC	0.23	0.38	0.78							
4	KCOV	0.36	0.59	0.52	0.74						
5	KSHR	0.15	0.25	0.21	0.60	0.75					
6	KSTR	0.11	0.40	0.33	0.44	0.42	0.77				
7	PDI	0.11	0.23	0.22	0.28	0.20	0.24	0.76			
8	PCI	0.15	0.37	0.28	0.39	0.23	0.44	0.37	0.80		
9	ADMIN	0.20	0.22	0.15	0.46	0.28	0.37	0.47	0.36	0.79	
10	MRIN	0.15	0.30	0.17	0.34	0.33	0.32	0.44	0.49	0.55	0.73

#### Table 2 Discriminant validity

Notes: The square root of AVE values is shown on the diagonal and printed in bold; off-diagonal elements are the construct variables correlations. OGCUL – organizational culture, LMS – leadership and management support, KAC – knowledge acquisition, KCOV – knowledge conversion, KSHR – knowledge sharing, KSTR – knowledge storage, PDI – product innovation, PCI – process innovation, ADMIN – administrative innovation, MRIN – marketing innovation, RMSEA – root mean square error of approximation, SRMR – standardized root mean square residual, NFI – normed fit index; NNFI – non-normed fit index; CFI – comparative fit index; IFI – incremental fit index; RFI – relative fit index; GFI – goodness of the fit index

Notes: The square root of AVE values is shown on the diagonal and printed in bold; off-diagonal elements are the construct variables correlations. OGCUL - organizational culture, LMS - leadership and management support, KAC - knowledge acquisition, KCOV - knowledge conversion, KSHR -knowledge sharing, KSTR - knowledge storage, PDI - product innovation, PCI - process innovation, ADMIN - administrative innovation, MRIN - marketing innovation, RMSEA - root mean square error of approximation, SRMR - standardized root mean square residual, NFI - normed fit index; NNFI - non-normed fit index; CFI - comparative fit index; IFI - incremental fit index; RFI - relative fit index; GFI -goodness of the fit index

The study further assessed the reliability and validity of the second-order innovation model. As presented in Figure A5 in Appendix 1, the value of CR exceeds the cutoff point value of 0.7, demonstrating internal consistency reliability satisfied with the model. The model fit indexes are RMSEA = 0.079, NFI = 0.92, CFI = 0.93, IFI = 0.93 and GFI = 0.90. These results were within the acceptable range, which supports the model's close fit to the data. Moreover, the path coefficients (correlations) between innovation (second-order constructs) and its constructs (first-order constructs) were strongly statistically significant. It suggests innovation can be best represented by the four constructs: PDI, PCI, ADMIN and MRIN.

The unidimensionality and multi-normality assumptions were also tested. We performed a crossloading analysis using confirmatory factor analysis to determine whether each manifest variable only loaded on one common factor. The results presented in Appendices 3 and 4 indicate the model has no cross-loading among the observed variables or error terms, suggesting the unidimensionality assumption is met. The study used two approaches to evaluate normality assumptions. The underlying univariate normality for each item is computed using skewness or kurtosis. The bivariate normality for a pair of items was assessed using the RMSEA. The result showed the values of skewness or kurtosis for all items were less than the recommended standard value of 3, and the RMSEA values for all pairs of items were in the acceptable range of less or equal to 0.1 as suggested by Joreskog (2005). Furthermore, the fitness of the models has been tested using multistatistics indices: the x2 values for the three models are statistically significant, indicating poor fit. As x2 is very sensitive to sample size, the model's poor fit is due to the small sample size (153 in the current study). Alternatively, normed (the ratio of x2 to the degree of freedom) for all models obtained an acceptable value, which is less than the cut-off point value of 5 claimed by Wheaton etal. (1977). The models also best fit the data with respect to all other indices, such as RMSEA, standardized root mean square residual (SRMR), normed fit index (NFI), non-normed fit index (NNFI), comparative fit index (CFI), incremental fit index (IFI), relative fit index (RFI) and GFI (Table 3).

# 4.2 Structural model

The study used SEM to test the hypotheses using the LISREL 8.80 SIMPLIS program with the maximum likelihood (ML) estimation technique. For the concern of model fit, which partially relies on the number of factors in a model (Anderson and Gerbing, 1988; Cheung and Rensvold, 2002; Hair et al., 2019), we separately run two models with respect to KM and innovation relationships.

Indices statistics	Model 1	Model 2	Model 3	Recommended cut-off rule	Reference
χ <sup>2</sup>	41.87	133.59	319.56	-	
Df	21	69	110	-	
<i>p</i> -value	0.004	0.000	0.000	> 0.05	
$\chi^2/df$	1.99	1.93	2.90	< 5	Wheaton <i>et al.</i> (1977)
RMSEA	0.08	0.078	0.068	< 0.1	Steiger and Lind (1980)
SRMR	-	0.062	0.045	< 0.08	Hu and Bentler (1999)
NEL	0.96	0.91	0.90	> 0.90	Bentler and Bonett (1980)
NNEL	-	0.94	0.90	> 0.90	Bentler and Bonett (1980)
CFI	0.98	0.95	0.93	> 0.90	Bentler (1990)
IFI	0.98	0.95	0.93	> 0.90	Bentler (1990)
GFI	0.94	0.90	0.90	≥ 0.90	Jöreskog and Sörbom (1989)

Table 3 Summar	y of model	fit indices
----------------	------------	-------------

Source: Adopted from Kalko et al. (2022)

In the first model, Figure 2, two constructs, KAC and KSHR, were assumed as endogenous latent variables, and all constructs of innovation were endogenous variables. The second model, Figure 3, used KCOV and KSTR as exogenous variables with all innovation constructs. The first task in structural model estimation is testing the goodness of fit of the model. For Model 1 (Figure 2), the fit indices are normed x2=2.87, RMSEA = 0.067, NFI = 0.91, CFI = 0.92, IFI = 0.92, GFI = 0.90. These results of model fit indexes are within the acceptable range, suggesting the hypothetical model best fits the data. Table 4 reports the detailed results of Models 1 and 2. After the model's goodness of fit was confirmed, we turned to examine the significance of each path coefficient. KAC has a significant positive effect on two constructs of innovation: PDI (unstandardized coefficient (UC) = 0.18, p-value = 0.000, f-value = 2.00) and PCI (UC = 0.24, p-value = 0.000, f-value = 2.74).



Figure 2 Effect of KM (KAC and KSHR) on innovation



Figure 3 Effect of KM (KCOV and KSTR) on innovation

The result implies the practice of acquiring and using useful knowledge positively contributes to innovation activity in a firm. The coefficients on ADMIN and MRIN were found insignificant.

The study finds a strong significant positive relationship between KSHR and all innovation constructs: PDI (UC = 0.21, p-value = 0.000, f-value = 2.25), PCI (UC = 0.23, p-value = 0.000, f-value = 2.52), ADMIN (UC = 0.30, p-value = 0.000, f-value = 3.15) and MRIN (UC = 0.36, p-value = 0.000, f-value = 3.69). The results support the hypothesis that knowledge-sharing behavior among employees or firms is a key driver of innovation.

The goodness of fit test for Model 2 has also yielded acceptable results in all fit indexes. The normed \*2=2.90, RMSEA = 0.073, NFI = 0.93, CFI = 0.94, IFI = 0.94 and GFI = 0.91. These results collectively suggest the conceptual model best fits the data. As presented in Table 4, KCOV has a potential

significant positive impact on all endogenous construct variables: PDI (UC = 0.25, p-value = 0.000, f-value = 2.23), PCI (UC = 0.27, p-value = 0.000, f-value = 2.27), ADMIN (UC = 0.39, p-value = 0.000, f-value = 3.66) and MRIN (UC = 0.29, p-value = 0.000, f-value = 2.71). The hypothesis that posits KCOV, which involves converting existing or new ideas or insights into explicit knowledge, is significantly attributed to innovation is confirmed. Similarly, KSTR has a significant positive relationship with constructs of innovation: PDI (UC = 0.23, p-value = 0.000, f-value = 2.05), PCI (UC = 0.40, p-value = 0.000, f-value = 3.83), ADMIN (UC = 0.26, p-value = 0.000, f-value = 2.50) and MRIN (UC = 0.32, p-value = 0.000, f-value = 2.50) and MRIN (UC = 0.32, p-value = 0.000, f-value = 2.89). The results provide evidence that storing knowledge in a database/manual/schema and reusing it for the decision-making process has a significant influence on a firm's innovation.

Variable	UC	Stand. error	t-value	Hypothesis			
Model 1: KM (KAC and KSHR) and innovation							
$KAC \rightarrow PDI$	0.18	0.092	2.00*	Supported			
$KSHR \rightarrow PDI$	0.21	0.095	2.25*	Supported			
$KAC \rightarrow PCI$	0.24	0.088	2.74**	Supported			
KSHR → PCI	0.23	0.090	2.52*	Supported			
KAC → ADMIN	0.08	0.090	0.84	Not supported			
$KSHR \rightarrow ADMIN$	0.30	0.096	3.15**	Supported			
$KAC \rightarrow MRIN$	0.11	0.090	1.23	Not supported			
$KSHR \rightarrow MRIN$	0.36	0.098	3.69**	Supported			
RMSEA = 0.067	NFI = 0.91	CFI = 0.92 IFI = 0.92	GFI = 0.90				
Model 2: KM (KCO)	/and KSTR) ar	nd innovation					
KCOV → PDI	0.25	0.11	2.23*	Supported			
$KSTR \rightarrow PDI$	0.23	0.11	2.05*	Supported			
KCOV → PCI	0.27	0.10	2.74**	Supported			
$KSTR \rightarrow PCI$	0.40	0.10	3.83**	Supported			
KCOV → ADMIN	0.39	0.11	3.66**	Supported			
$KSTR \to ADMIN$	0.26	0.10	2.50*	Supported			
$KCOV \rightarrow MRIN$	0.29	0.11	2.71**	Supported			
$KSTR \rightarrow MRIN$	0.32	0.11	2.89**	Supported			
RMSEA = 0.073	NFI = 0.93	CFI = 0.94 IFI = 0.94	GFI = 0.91				
Notes: Where: KAC – knowledge acquisition, KSHR – knowledge sharing, KCOV – knowledg conversion, KSTR – knowledge storage, PDI – product innovation, PCI – process innovatior							

#### Table 4 Summary of structural model results

Notes: Where: KAC – knowledge acquisition, KSHH – knowledge sharing, KCOV – knowledge conversion, KSTR – knowledge storage, PDI – product innovation, PCI – process innovation, ADMIN – administrative innovation, MRIN – marketing innovation, RMSEA – root mean square error of approximation, NFI – normed fit index, CFI – comparative fit index, IFI – incremental fit index, GFI – goodness-of-fit index, unstandardized coefficient (UC) represents path coefficient, KAC, KSHR, KCOV and KSTR are KSI (exogenous latent) variables. PDI, PCI, ADMIN and MRIN are ETA (endogenous latent) variables. Note: \*\*,\* - indicates a significant level at 1 and 5%, respectively

Notes: Where: KAC - knowledge acquisition, KSHR - knowledge sharing, KCOV - knowledge conversion, KSTR - knowledge storage, PDI - product innovation, PCI - process innovation, ADMIN - administrative innovation, MRIN - marketing innovation, RMSEA - root mean square error of approximation, NFI - normed fit index, CFI - comparative fit index, IFI - incremental fit index, GFI -goodness-of-fit index, unstandardized coefficient (UC) represents path coefficient, KAC, KSHR, KCOV and KSTR are KSI (exogenous latent) variables. PDI, PCI, ADMIN and MRIN are ETA (endogenous latent) variables. Note: \*\*,\* - indicates a significant level at 1 and 5%, respectively

The study has also done further analysis to examine the relationship between organizational factors (OGCUL and LMS) and KM. In an extension of this model, a mediation analysis has been performed where OGCUL and/or LMS are assumed as an exogenous independent variable. KM is used as a mediate variable, whereas innovation is the endogenous dependent variable. For simplicity of the model, a single mediating variable from the KM construct, KSHR, was only introduced. Two mediation models were separately run. One is OGCUL (OGCUL), and the other is LMS for the model fit concern. In Figure 4, four constructs of KM were regressed on OGCUL, and the coefficients of all endogenous

constructs were significant and positive. Table 5 presents the detailed results on KAC (UC = 0.28, p-value = 0.000, t-value = 3.02), KCOV (UC = 0.41, p-value = 0.000, t-value = 3.95), KSHR (UC = 0.23, p-value = 0.000, t-value = 2.34) and KSTR (UC = 0.22, p-value = 0.000, ř-value = 2.07). It demonstrates that OGCUL is an important driver of KM.

The first mediation model examines the impact of OGCUL on innovation through KSHR. Following the common procedure of mediation analysis (Baron and Kenny, 1986), we first tested whether OGCUL, an independent variable, significantly impacts innovation, a dependent variable, without taking a mediate variable. The results in Figure 5 showed that there is a significant positive relationship between OGCUL and all innovation constructs: PDI (UC = 0.15, p-value = 0.000, t-value = 1.97), PCI (UC = 0.20, p-value = 0.000, t-value = 2.26), ADMIN (UC = 0.23, p-value = 0.000, t-value = 2.45) and MRIN (UC = 0.21, p-value = 0.000, t-value = 2.37). By this result, the first condition (the relationship between dependent and independent variables should be significant) is met. The second condition that demands a significant relationship between mediating and independent variables is shown in Figure 4.



Figure 4 Effectof OGCULon KM

At the third stage, a model with a mediate variable, KSHR, was run in the LISREL 8.80 SIMPLIS program with an ML estimation technique, and the results were summarized in Figure 6 and Table 5.

The goodness of fit model test has yielded acceptable results in all fit indexes. These include normed \*2=2.31, RMSEA = 0.073, NFI = 0.92, CFI = 93, I FI = 0.93 and GFI = 0.90, suggesting the hypothetical model fits the data well. It is indicated in Table 5 that OGCUL has a significant positive direct impact on the mediate variable, KSHR (UC = 0.21, p-value = 0.000, Lvalue = 1.99). However, no direct significant coefficient is found between OGCUL and dependent variables, innovation constructs. Similarly, the indirect effect of OGCUL on all dependent variables is insignificant.

Moreover, the cumulative effects of OGCUL on PCI (UC = 0.18, p-value = 0.000, Lvalue = 2.08) and ADMIN (UC = 0.21, p-value = 0.000, Lvalue = 2.32) are significant. We recalled the conditions proposed by Baron and Kenny (1986) for mediation analysis to sum up the results. The first condition that requires there to be a significant relationship between dependent and independent variables was satisfied (Figure 5). Second, the mediate variable should significantly relate to the independent variable. It was met in this study (Figure 4). The third condition requires a significant relationship between a dependent and the mediate variable; it is also confirmed (Figure 3). The present study runs three equations separately to test the conditions. Fourth, the parameter coefficient between OGCUL and dependent variables (i.e. PDI, PCI, ADMIN and MRIN) in Figure 6, the mediation diagram, may be significant or insignificant, but it should be smaller than the parameter coefficient in Figure 5. If the first three conditions are met and the coefficient in Figure 6 is significant, partial mediation is assumed. Otherwise, complete mediation is concluded. Thus, in this study, complete mediation is inferred.

We also run a separate model to examine the effect of LMS on KM, as indicated in Figure 7 and Table 6. The model has yielded an acceptable fit index, which involves normed  $x^2 = 2.24$ , RMSEA = 0.091, NFI = 0.91, CFI = 92, IFI = 0.92 and GFI = 0.90.

Variable	UC	Std. error	t-value	Hypothesis
Model 1: the effect of orga	nization culture on KM			
OGCUL → KAC	0.28	0.093	3.02**	Supported
OGCUL → KCOV	0.41	0.10	3.95**	Supported
OGCUL → KSHR	0.23	0.099	2.34*	Supported
OGCUL → KSTR	0.22	0.11	2.07*	Supported
RMSEA = 0.080 NFI = 0.92	2 CFI = 0.94 IFI = 0.94 GFI = 0	.91		
Model 2: the effect of orga	nization culture on innovation			
OGCUL → PDI	0.15	0.076	1.97*	Supported
OGCUL → PCI	0.20	0.089	2.26*	Supported
OGCUL → ADMIN	0.23	0.094	2.45*	Supported
OGCUL → MRIN	0.21	0.089	2.37*	Supported
RMSEA = 0.081 NFI = 0.95	5 CFI = 0.96 IFI = 0.96 GFI = 0	.90		
Model 3: the effect of orga	nization culture on innovation.	the mediation of KSHR		
Variable	Total effects	Direct effects	Indirect effects	
OGCUL → KSHR	0.21 (1.99*)	0.21(1.99*)	0.00	
OGCUL → PDI	0.15(1.61)	0.081 (0.91)	0.06 (1.73)	
OGCUL → PCI	0.18 (2.08*)	0.11(1.32)	0.07 (1.78)	
OGCUL → ADMIN	0.21 (2.32*)	0.12(1.66)	0.07 (1.77)	
OGCUL → MRIN	0.15 (1.67)	0.061 (0.71)	0.09(1.84)	
KSHR → PDI	0.30 (2.31*)	0.30 (3.16**)	0.00	
KSHR → PCI	0.33 (2.24*)	0.33 (3.52**)	0.00	
KSHR → ADMIN	0.28 (2.67*)	0.28 (3.36**)	0.00	
$KSHR \rightarrow MRIN$	0.43 (3.34**)	0.43 (4.18**)	0.00	
RMSEA = 0.073 NEL = 0.92	CEL = 0.93 IEL = 0.93 GEL = 0	90		

Table 5 Summary of structural model results

Note: Where: LMS – leadership and management support, KSHR – knowledge sharing, PDI – product innovation, PCI – process innovation, ADMIN – administrative innovation, MRIN – marketing innovation, RMSEA – root mean square error of approximation, NFI – normed fit index, CFI – comparative fit index, IFI – incremental fit index, GFI – goodness-of-fit index, the value in bracket is *t*-value, OGCUL is KSI (exogenous latent) variable, KSHR, PDI, PCI, ADMIN and MRIN are ETA (endogenous latent) variables, coefficients are computed by ML estimation method. Note: the coefficient of indirect effect is computed by OGCUL direct effect on KSHR coefficient (i.e. 0.21) times KSHR direct effects on innovation construct variables. Total effect = direct effect + indirect effect. \*\*,\* - indicate a significant level at 1 and 5%, respectively

The result in Table 6 shows that LMS has a strong significant positive impact on all dimensions of KM: KAC (UC = 0.46, p-value = 0.000, f-value = 4.97), KCOV (UC = 0.64, p-value = 0.000, f-value = 5.73), KSHR (UC = 0.36, p-value = 0.000, f-value = 3.53) and KSTR (UC = 0.47, p-value = 0.000, f-value = 4.68). It supports the hypothesis that LMS strengthens/improve KM practice in a firm. In addition, OGCUL LMS are also important drivers of KM.

Furthermore, this study has examined how LMS impact innovation through KSHR intervention. Twofold procedures were followed to do so. The first procedure was regressing innovation, dependent variables on LMS, independent variable. Figure 8 and Table 6 present the detailed results. It is clearly seen that LMS has a significant positive linkage with three dependent variables: PCI (UC = 0.30, p-value = 0.000, f-value = 3.43), ADMIN (UC = 0.30, p-value = 0.000, f-value = 3.32) and MRIN (UC = 0.24, pvalue = 0.000, f-value = 2.59). The result suggests that firm LMS has a significant role in innovation practices. After assuring that Conditions 1 (Figure 8) and 2 (Figure 7) were met, we turned to run the model with a single mediate variable, KSHR. Figure 9 and Table 6 provide the detailed results. The model goodness of fit is normed x2=2.75, RMSEA = 0.077, NFI = 0.94, CFI = 94, IFI = 0.93 and GFI = 0.91. These results indicate the model fits the data well. In Figure 9, the parameter coefficients can be interpreted as direct effects, represented by the path lines pointing to the endogenous latent (ETA) variables (KSHR, PDI, PCI, ADMIN, MRIN) and indirect effects, defined by the path line going to the dependent variables (PDI, PCI, ADMIN and MRIN) through KSHR, mediating variable.



Figure 5 Effect of OGCUL on innovation

With regard to direct effects, the coefficient on the mediate variable (KSHR) (UC = 0.22, t-value = 2.29), PCI (UC = 0.29, t-value = 3.32) and ADMIN (UC = 0.26, t-value = 2.83) showed significant positive. On the other hand, LMS has an indirect significant positive impact on MRIN (UC = 0.08, t-value = 1.99). Overall results indicate a significant positive coefficient for three dependent variables: PCI (UC = 0.33, t-value = 3.71), ADMIN (UC = 0.31, t-value = 3.40) and MRIN (UC = 0.24, t-value = 2.54). In this mediation model, partial mediation is assumed for PCI and ADMIN, while complete mediation is observed for MRIN. The first condition for mediation analysis is not satisfactory for PDI; therefore, mediation analysis does not work well for the variable. It can be concluded that KSHR would mediate the analysis of organizational factors and innovation relationships. KSHR has a significant positive effect on all constructs of innovation.

# 5. Discussion

This study found that KAC appeared to have a significant positive impact on PDI and PCI. The result reveals that the extent to which a firm learns and acquires knowledge from different parts (Huber, 1991) has played a significant role in firm innovation activity. However, a firm's capacity to implement acquired knowledge into the production system is essential and decides the maximum benefits to be generated. In addition, KSHR appeared to be the key driving factor for all constructs of innovation, such as PDI, PCI, ADMIN and MRIN. It suggests the culture of KSHR among employees or between firms in the same industry highly encourages innovative activities. This view supports the perspective of the dynamic knowledge creation theory (Nonaka, 1994) and empirical findings (du Plessis, 2007; Lin, 2007; Wang and Noe, 2010; Lee and Choi, 2003; Gomezelj etal., 2011; Han and Chen, 2018) that confirm KM processes (such as KSHR, KSTR and KAC) encourage innovation. KCOV has a strong positive relationship with PDI, PCI, ADMIN and MRIN.



Figure 6 Effect of OGCUL on innovation: the mediating role of KSHR



Figure 7 Effect of LMSon KM

#### Table 6 Summary of structural model results

Variable		UC		Std. error	t-value	Hypothesis
Model 1: LMS and	d KM					
$LMS \rightarrow KAC$		0.46		0.093	4.97**	Supported
LMS → KCOV		0.64		0.11	5.73**	Supported
$LMS \rightarrow KSHR$		0.36		0.10	3.53**	Supported
$LMS \rightarrow KSTR$		0.47		0.10	4.68**	Supported
RMSEA = 0.091	NFI = 0.91	CFI = 0.92	IFI = 0.92	GFI = 0.90		
Model 2: LMS and	d innovation					
$LMS \rightarrow PDI$		0.15		0.092	1.64	Not supported
$LMS \rightarrow PCI$		0.30		0.088	3.43**	Supported
LMS ADMIN		0.30		0.092	3.32**	Supported
$LMS \rightarrow MRIN$		0.24		0.093	2.59**	Supported
RMSEA = 0.073	NFI = 0.93	CFI = 0.94	IFI = 0.94	GFI = 0.91		
Model 3: LMS and	d innovation:	with mediati	on of KSHR			
Variable		Total effe	cts	Direct effects	Indirect effects	
$LMS \rightarrow KSHR$		0.22 (2.29	)*)	0.22 (2.29*)	0.00	
$LMS \rightarrow PDI$		0.15 (1.58	3)	0.096 (1.02)	0.05 (1.69)	
$LMS \rightarrow PCI$		0.33(3.71	**)	0.29 (3.23**)	0.04 (1.68)	
LMS> ADMIN		0.31 (3.40	)**)	0.26 (2.83**)	0.06 (1.83)	
$LMS \rightarrow MRIN$		0.24(2.54	*)	0.16 (1.79)	0.08 (1.99*)	
$KSHR \rightarrow PDI$		0.23 (2.31	I*)	0.23 (2.31*)	0.00	
$KSHR \rightarrow PCI$		0.20 (2.24	<sup>1*</sup> )	0.20 (2.24*)	0 0.00	
$KSHR\toADMIN$		0.26 (2.67	7*)	0.26 (2.67**)	0.00	
$KSHR \rightarrow MRIN$		0.34 (3.34	l**)	0.34 (3.34**)	0.00	
RMSEA = 0.077	NFI = 0.94	CFI = 0.94	IFI =0.93	GFI = 0.91		

Notes: Where: LMS – leadership and management support, KSHR – knowledge sharing, PDI – product innovation, PCI – process innovation, ADMIN – administrative innovation, MRIN – marketing innovation, RMSEA – root mean square error of approximation, NFI – normed fit index, CFI – comparative fit index, IFI – incremental fit index, GFI – goodness-of-fit index, the value in bracket is t-value, LMS is KSI (exogenous latent) variable, KSHR, PDI, PCI, ADMIN and MRIN are ETA (endogenous latent) variables, coefficients are computed by ML estimation method. Note: the coefficient of indirect effect is computed by LMS direct effect on KSHR coefficient (i.e. 0.22) times KSHR direct effects on innovation construct variables. Total effect = direct effect + indirect effect. \*\*,\* - indicate a significant level at 1 and 5%, respectively

It suggests that a firm with a higher practice of knowledge conversation (from tacit to explicit, or tacit to tacit, or explicit to explicit or explicit to tacit knowledge) is more likely to be innovative in all dimensions. As tacit knowledge is an abstract that dwells in individuals' minds, it requires reasonable experience to transform it into explicit knowledge. The result supports the theory of organizational knowledge creation (Nonaka, 1994) and is consistent with prior studies (Byosiere and Luethge, 2008; Lee et al., 2005) that have shown KCOV is an essential driver of innovation.

The findings of this study also show that organizational factors (i.e. OGCUL and LMS) have a strong significant relationship with KM dimensions. It suggests that the success of the KM process highly depends on an organization's culture, which entails a sense of security, a lack of fear, openness, trust and transparency. Moreover, good LMS create a conducive environment for KM practices such as employee interaction, which increases the culture of sharing, learning and creating new ideas. It also makes resources available for a KM strategy that leads firms to be knowledge-intensive and enhances the quality of the decision-making process. In addition, KSHR significantly mediates the relationship between organizational factors (both OGCUL and LMS) and innovation, suggesting organizational factors are a valuable source of KM and, subsequently, innovation. Our study provides consistent results with previous studies that have found organizational factors encourage dialog between employees or groups, which is the basis for knowledge creation and innovation (De Long and Fahey, 2000; McDermott and O'Dell, 2001; Lee et al., 2012; Lee and Choi, 2003; Hsu et al., 2014; Wang et al., 2011; De Long and Fahey, 2000; Kog et al., 2019; Lin, 2007; Lee et al., 2012; Noruzy et al., 2013; Lee and Choi, 2003; Chang et al., 2015). Overall, the results confirm the theoretical view that knowledge is an asset that resides in individuals' minds, groups or organizational processes.





The role of KM is multi-faceted. For example, it initiates knowledge-based action, increases competitive advantage, creates an innovative culture and improves decision-making, customer service and productivity. Besides this, it is essential to understand the types and sources of knowledge to well link them to corporate core values. In this respect, according to De Long and Fahey, 2000, knowledge can appear in three modes:

- 1. human knowledge that refers to individual know-how or knows howto do;
- 2. social or collective knowledge-exists in social interaction, e.g. in research teams; and
- 3. structural knowledge-exists in a system, tools or process of an organization.

It can be concluded from this view that the effectiveness of KM relies on the firm's capacity to identify how and where knowledge can be obtained and integrated with corporate strategy.

# 5.1 Theoretical contributions

Theoretically, the current research extends and refines leadership, KM and the innovation literature in a variety of ways: first, the findings of this study provide empirical support for integrating the dynamic KM capability perspective (Gold et al., 2001) and the knowledge-based view (Kogut and Zander, 1992; Grant, 1996) by underscoring the mediating role of KSHR in the positive linkage between organizational factors (i.e. OGCUL and LMS) and organizational innovation.



Figure 9 Effect of LMS on innovation: the mediating role of KSHR

Second, many scholars have indicated that organizational factors such as OGCUL, leadership and top management support are crucial antecedents of a company's ability to innovate. Many authors have recently explored the impact of various organizational factors on KM and various organizational outcomes. Despite the growing research interest in organizational factors, KM and innovation, only a few studies have presented empirical evidence linking various organizational factors and KM processes

to innovation effectiveness. Furthermore, our results prove the complete mediating role of KM in the relationship between organizational factors and innovation. Therefore, our study contributes to the current growing literature on direct effects, indirect effects and total effect analysis (Baron and Kenny, 1986; MacKinnon etal., 2012).

# 5.2 Practical implications

The findings of this study would serve as a guide for policy-makers and managers of manufacturing firms in developing countries in the formulation of policies and long-term strategies. It may also provide a better understanding of the causal relationship between organizational factors, KM and innovation, which in turn has value to directors and managers in manufacturing firms in developing countries as a reference for building a good OGCUL, serving as practical guidance for effective leadership and providing organizational or management support. Specifically, our findings would have the following practical implications: first, firms need to have a combination of KM processes (such as acquisition, storage, sharing and conversion). In practice, developing countries such as Ethiopia have based their innovation strategy on knowledge and technology acquisition through encouraging foreign direct investment. It is not in doubt that Ethiopia has been benefiting from the strategy as a lot of foreign companies have opened their subsidiaries in the country. However, in our view, more emphasis on KAC strategy would not take a firm a long time to sustain its innovative activity because it is likely available to firms operating in the same industry, as well as it may hurt a firm's competitive advantage. In addition, by its nature, knowledge may not be retained for future use; it may expire soon.

Second, the current highly impulsive and rapid change in the business environment changes the way firms have to operate and deliver products or services. Knowledge (both tacit and explicit) is a resource that can provide a competitive advantage if used well for the intended purpose. In real practice, firms often face challenges in determining where to get knowledge from and how to value or manage it. Besides, knowledge can be obtained from three sources:

- 1. knowledge can exist in individuals' minds (skills, experience, ideas and insight);
- 2. knowledge can dwell in a group, which we can call collective knowledge (a team of scientists or researchers); and
- 3. knowledge can be embodied in an organization's systems, tools, procedures, policies, etc.

Knowledge cannot be a valuable resource unless it is obtained and used in designing or producing a product or service. To integrate knowledge with business strategies, there should be a platform or framework that helps to manage it properly. Firm managers, policymakers and other concerned bodies would consider the three sources of knowledge to foster innovative activities and obtain a competitive advantage. In addition, we recommend more emphasis be placed on firm-specific factors (such as OGCUL, LMS and KM) to enhance the innovative capacity of a firm.

Finally, the most critical issue to be raised while designing an innovation strategy would be employees' willingness and passion to collaborate with others to develop new ideas, share ideas or implement policies. As knowledge resides in individuals' minds, the knowledge holder should have a passion to share it with those working with him or her. In practice, KSHR depends extremely on the passion and voluntariness of the two parts: knowledge provider and receiver. Therefore, firm managers would design a platform on how to motivate individuals to share their skills, experience and ideas with others by providing incentive packages, punishment and commitment. In this regard, we believe that our results would help individuals who are in the position to manage or regulate the manufacturing sector

in designing innovation policies, KM policies or technology management policies and business strategies.

# 5.3 Limitations and direction for future research

This study has three potential limitations: first, this study is based on a cross-sectional research design. Future research should include longitudinal design to get in-depth insights into the causal inferences. Second, only a few Ethiopian medium- and large-scale manufacturing firms were included in the sample. As a suggestion for future research, other researchers can include small-scale enterprises using large sample sizes and should examine the effects of organizational factors, KM and innovation across different industries. Finally, this study has only focused on investigating the mediating role of KSHR between organizational factors and innovation. Future research should test the mediating role of the KM process and its constituents (KAC, KCOV, KSHR and KSTR) between organizational factors and specific aspects of innovation to gain a full understanding of the critical role of KM in organizational innovation.

# 6. Conclusion

Knowledge is an intangible organizational resource that could provide a competitive advantage if properly acquired and used in a production process. Like other tangible resources, knowledge needs to be managed to get the most use out of it. KCOV, KAC, KSHR and KSTR are valuable dimensions of KM. The result gives a basis to conclude that a combination of KCOV, KSHR, KAC and KSTR would enable firms to become knowledgeintensive, which improves the quality of decisions and their ability to solve problems. Furthermore, organizational factors (i.e. OGCUL and LMS) appeared to be the essential enablers of KM. Where OGCUL encourages sharing ideas with passion among employees, more innovative ideas or knowledge will emerge and be implemented. LMS also play a vital role in encouraging KSHR, which in turn enhances innovation. On the other hand, leadership is a device that nurtures motivation, skills and competence for employees and fosters the successful generation and implementation of knowledge. From this perspective, we can infer that a KM system cannot function well without a good OGCUL and leadership and top management support.

# References

Adelekan, S.A. (2016), "The impact of organizational culture on innovation capability of SMEs: a case study of SMEs in Alimosho and Ojo local government area of Lagos state, Nigeria", International Journal of Economics, Commerce and Management, Vol. 4 No. 9, pp. 158-181.

Ajmal, M.M. and Koskinen, K.U. (2008), "Knowledge transfer in project-based organizations: an organizational culture perspective", Project Management Journal, Vol. 39 No. 1, pp. 7-15, doi: 10.1002/pmj.20031.

Alavi, M. and Leidner, D.E. (2001), "Review: knowledge management and knowledge management systems: conceptual foundations and research issues", MIS Quarterly, Vol. 25 No. 1, pp. 107-136, doi: 10.2307/3250961.

Alhady, M., Idris, A., Sawal, M., Azmi, N. and Zakaria, Z. (2011), "Knowledge sharing behavior and individual factors: a relationship study in the i-Class environment", In Proceeding of the International Conference on Management and Artificial Intelligence.

Al-Sa'di, A.F., Abdallah, A.B. and Dahiyat, S.E. (2017), "The mediating role of product and process innovations on the relationship between knowledge management and operational performance in manufacturing companies in Jordan", Business Process Management Journal, Vol. 23 No. 2, pp. 349-376, doi: 10.1108/BPMJ-03-2016-0047

Anderson, J.C. and Gerbing, D.W. (1988), "Structural equation modeling in practice: a review and recommended two-step approach", Psychological Bulletin, Vol. 103 No. 3, pp. 411-423.

Andreeva, T. and Kianto, A. (2011). Knowledge processes, knowledge-intensity and innovation: a moderated mediation analysis. Journal of Knowledge Management, Vol. 15 No. 6, pp. 1016-1034, doi: 10.1108/13673271111179343.

Anderson, J.C. and Gerbing, D.W. (1984), "The effect of sampling error on convergence, improper solutions, and goodness-of-fit indices for maximum likelihood confirmatory factor analysis", Psychometrika, Vol. 49 No. 2, pp. 155-173, doi: 10.1007/bf02294170.

Andries, P., De Winne, S. and Bos-Nehles, A. (2019), "Knowledge management practices for stimulating incremental and radical product innovation", in Tuncdogan, A., Lindgreen, A., Volberda, H. and van den Bosch, F. (Eds), Strategic Renewal: core Concepts, Antecedents, and Micro Foundations, Routledge, New York, NY.

Bagozzi, R. and Yi, Y. (1988), "On the evaluation of structural equation models", Journal of the Academy of Marketing Science, Vol. 16 No. 1, pp. 74-94, doi: 10.1007/bf02723327.

Bagozzi, R., Yi, Y. and Phillips, L. (1991), "Assessing construct validity in organizational research", Administrative Science Quarterly, Vol. 36 No. 3, pp. 421-458, doi: 10.2307/2393203.

Barney, J.B. (1986), "Organizational culture: can it be a source of sustained competitive advantage?", Academy of Management Review, Vol. 11 No. 3, pp. 656-665, doi: 10.5465/AMR.1986.4306261.

Baron, R. and Kenny, D. (1986), "The moderator-mediator variable distinction in social psychological research: conceptual, strategic, and statistical considerations", Journal of Personality and Social Psychology, Vol. 51 No. 6, pp. 1173-1182.

Bass, B. and Riggio, R. (2012), Transformational Leadership (2nd Edition), Lawrence Erlbaum Associates.

Bavik, Y.L., Tang, P.M., Shao, R. and Lam, L.W. (2018), "Ethical leadership and employee knowledge sharing: exploring dual-mediation paths", The Leadership Quarterly, Vol. 29 No. 2, pp. 322-332, doi: 10.1016/j.leaqua.2017.05.006.

Bentler, P.M. (1990), "Comparative fit indexes in structural models", Psychological Bulletin, Vol. 107 No. 2, pp. 238-246, doi: 10.1037/0033-2909.107.2.238.

Bentler, P. and Bonett, D. (1980), "Significance tests and goodness of fit in the analysis of covariance structures", Psychological Bulletin, Vol. 88 No. 3, pp. 588-606, doi: 10.1037/0033-2909.88.3.588.

Bollen, K.A. (1987), "Outliers and improper solutions", Sociological Methods & Research, Vol. 15 No. 4, pp. 375-384, doi: 10.1177/0049124187015004002.

Bollen, K. and Lennox, R. (1991), "Conventional wisdom on measurement: a structural equation perspective", Psychological Bulletin, Vol. 110 No. 2, pp. 305-314, doi: 10.1037/0033-2909.110.2.305.

Browne, M.W. and Cudeck, R. (1989), "Single sample cross-validation indices for covariance structures", Multivariate Behavioral Research, Vol. 24 No. 4, pp. 445-455, doi: 10.1207/s15327906 mbr2404\_4.

Byosiere, P. and Luethge, D.J. (2008), "Knowledge domains and knowledge conversion: an empirical investigation", Journal of Knowledge Management, Vol. 12 No. 2, pp. 67-78, doi: 10.1108/ 13673270810859523.

Carayannis, E.G., Gonzalez, E. and Wetter, J. (2003), "The nature and dynamics of discontinuous and disruptive innovations from a learning and knowledge management perspective", The International Handbook on Innovation, Elsevier Science.

Central Statistical Agency (2014), Report on manufacturing sectors.

Chang, J., Bai, X. and Li, J.J. (2015), "The influence of leadership on product and process innovations in China: the contingent role of knowledge acquisition capability", Industrial Marketing Management, Vol. 50, pp. 18-29, doi: 10.1016/j.indmarman.2015.04.014.

Chang, S. and Lee, M. (2008). "The linkage between knowledge accumulation capability and organizational innovation", Journal of Knowledge Management, Vol. 12 No. 1, pp. 3-20, doi: 10.1108/ 13673270810852359.

Chang, W.-J., Liao, S.-H. and Wu, T.-T. (2017), "Relationships among organizational culture, knowledge sharing, and innovation capability: a case of the automobile industry in Taiwan", Knowledge Management Research & Practice, Vol. 15 No. 3, pp. 471-490, doi: 10.1057/s41275-016-0042-6.

Chatzoglou, P. and Chatzoudes, D. (2018), "The role of innovation in building competitive advantages: an empirical investigation", European Journal of Innovation Management, Vol. 21, No 1, pp. 44-69, doi: 10.1108/EJIM-02-2017-0015.

Cheung, G.W. and Rensvold, R.B. (2002), "Evaluating goodness-of-fit indexes for testing measurement invariance", Structural Equation Modeling: A Multidisciplinary Journal, Vol. 9 No. 2, pp. 233-255, doi: 10.1207/s15328007sem0902\_5.

Chin, T.S., Kavitha, R., Yeow, J.A. and Eze, U.C. (2014), "The contributing roles of emotional intelligence and spiritual intelligence in entrepreneurial innovation and creativity", World Journal of Management, Vol. 5 No. 2, pp. 66-77.

Cho, T. and Korte, R. (2014), "Managing knowledge performance: testing the components of a knowledge management system on organizational performance", Asia Pacific Education Review, Vol. 15 No. 2, pp. 313-327, doi: 10.1007/s12564-014-9333-x.

Cho, H.-J. and Pucik, V. (2005), "Relationship between innovativeness, quality, growth, profitability, and market value", Strategic Management Journal, Vol. 26 No. 6, pp. 555-575, doi: 10.1002/smj.461.

Choi, S.B., Kim, K., Ullah, S.E. and Kang, S.-W. (2016), "How transformational leadership facilitates innovative behavior of Korean workers: examining mediating and moderating processes", Personnel Review, Vol. 45 No. 3, pp. 459-479, doi: 10.1108/PR-03-2014-0058.

Connelly, C.E. and Kelloway, E.K. (2003), "Predictors of employees' perceptions of knowledge sharing cultures", Leadership & Organization Development Journal, Vol. 24 No. 5, pp. 294-301, doi: 10.1108/01437730310485815.

Crawford, C.B. (2005), "Effects of transformational leadership and organizational position on knowledge management", Journal of Knowledge Management, Vol. 9 No. 6, pp. 6-16, doi: 10.1108/ 13673270510629927.

Daksa, M.D., Yismaw, M.A., Lemessa, S.D. and Hundie, S.K. (2018), "Enterprise innovation in developing countries: evidence from Ethiopia", Journal of Innovation and Entrepreneurship, Vol. 7 No. 1, pp. 1-19, doi: 10.1186/s13731-018-0085-4.

Darroch, J. (2005), "Knowledge management, innovation and firm performance", Journal of Knowledge Management, Vol. 9 No. 3, pp. 101-115, doi: 10.1108/13673270510602809.

Darroch, J. and McNaughton, R. (2002), "Examining the link between knowledge management practices and types of innovation", Journal of Intellectual Capital, Vol. 3 No. 3, pp. 210-222, doi: 10.1108/14691930210435570.

Davenport, T.H., De Long, D.W. and Beers, M.C. (1998), "Successful knowledge management projects", Sloan Management Review, Vol. 39 No. 2, pp. 43-57.

De Jong, J.P.J. and Den Hartog, D.N. (2007), "How leaders influence employees' innovative behavior", European Journal of Innovation Management, Vol. 10 No. 1, pp. 41-64, doi: 10.1108/ 14601060710720546.

De Long, D.W. and Fahey, L. (2000), "Diagnosing cultural barriers to knowledge management", Academy of Management Perspectives, Vol. 14 No. 4, pp. 113-127, doi: 10.5465/AME.2000.3979820.

Do, H., Mazzarol, T., Soutar, G.N., Volery, T. and Reboud, S. (2018), "Organisational factors, anticipated rents and commercialization in SMEs", International Journal of Innovation Management, Vol. 22 No. 2, pp. 1-30, doi: 10.1142/S1363919618500184.

Donate, M.J. and Guadamillas, F. (2011), "Organizational factors to support knowledge management and innovation", Journal of Knowledge Management, Vol. 15 No. 6, pp. 890-914, doi: 10.1108/ 13673271111179271.

Donnelly, R. (2018), "Aligning knowledge sharing interventions with the promotion of firm success: the need for SHRM to balance tensions and challenges", Journal of Business Research, doi: 10.1016/j. jbusres.2018.02.007.

du Plessis, M. (2007), "The role of knowledge management in innovation", Journal of Knowledge Management, Vol. 11 No. 4, pp. 20-29, doi: 10.1108/13673270710762684.

Erena, O.T., Kalko, M.M. and Debele, S.A. (2021), "Technical efficiency, technological progress and productivity growth of large and medium manufacturing industries in Ethiopia: a data envelopment analysis", Cogent Economics & Flnance, Vol. 9No. 1, pp. 1-38, doi: 10.1080/23322039.2021.1997160.

Erena, O.T., Kalko, M.M. and Debele, S.A. (2022), "Corporate governance mechanisms and firm performance: empirical evidence from medium and large-scale manufacturing firms in Ethiopia", Corporate Governance: The International Journal of Business in Society, Vol. 22 No. 2, pp. 213-242, doi: 10.1108/CG-11 -2020-0527.

Fan, X., Thompson, B. and Wang, L. (1999), "Effects of sample size, estimation methods, and model specification on structural equation modeling fit indexes", Structural Equation Modeling: A Multidisciplinary Journal, Vol. 6 No. 1, pp. 56-83, doi: 10.1080/10705519909540119.

Fornell, C. and Larcker, D.F. (1981), "Evaluating structural equation models with unobservable variables and measurement error", Journal of Marketing Research, Vol. 18No. 1, pp. 39-50, doi: 10.1177/002224378101800104.

Freeman, C. (1995), "Innovation and growth", Chapters, in Dodgson, M. and Rothwell, R. (Eds), The Handbook of Industrial Innovation, chapter 7, Edward Elgar Publishing.

Furman, J.L., Porter, M.E. and Stern, S. (2002), "The determinants of national innovative capacity", Research Policy, Vol. 31 No. 6, pp. 899-933, doi: 10.1016/s0048-7333(01)00152-4.

Gloet, M. and Terziovski, M. (2004), "Exploring the relationship between knowledge management practices and innovation performance", Journal of Manufacturing Technology Management, Vol. 15 No. 5, pp. 402-409, doi: 10.1108/17410380410540390.

Gold, A.H., Malhotra, A. and Segars, A.H. (2001), "Knowledge management: an organizational capabilities perspective", Journal of Management Information Systems, Vol. 18 No. 1, pp. 185-214, doi: 10.1080/07421222.2001.11045669.

Gomezelj, O.D., Antončič, B. and Ruzzier, M. (2011), "Developing and testing a multi-dimensional knowledge management model on Slovenian SMEs", Baltic Journal of Management, Vol. 6 No. 2, pp. 179-204, doi: 10.1108/17465261111131802.

Grant, R.M. (1996), "Toward a knowledge-based theory of the firm", Strategic Management Journal, Vol. 17 No. 2, pp. 109-122, doi: 10.1002/smj.4250171110.

Grant, S.B. (2016), "Classifying emerging knowledge sharing practices and some insights into antecedents to social networking: a case in insurance", Journal of Knowledge Management, Vol. 20 No. 5, pp. 898-917, doi: 10.1108/JKM-11-2015-0432.

Guaderrama, A.I.M. (2016), "Culture and affective commitment: do they influence the internal knowledge transfer?", Contadun'a y Administración, Vol. 61 No. 4, pp. 666-681, doi: 10.1016/j. cya.2016.06.003.

Gupta, A.K. and Govindarajan, V. (2000), "Knowledge management's social dimension: lesson from nucor steel", MIT Sloan Management Review, Vol. 42 No. 1, pp. 71-80.

Hair, J.F., Black, W.C., Babin, B.J. and Anderson, R.E. (2019), Multivariate Data Analysis, 8th ed., Cengage Learning EMEA, Hampshire.

Han, Y. and Chen, G. (2018), "The relationship between knowledge sharing capability and innovation performance within industrial clusters: evidence from China", Journal of Chinese Economic and Foreign Trade Studies, Vol. 11 No. 1, pp. 32-48, doi: 10.1108/JCEFTS-06-2017-0018.

Hashi, I. and Stojcic, N. (2010), "The impact of innovation activities on firm performance using a multistage model: evidence from the community innovation survey 4", doi: 10.2139/ssrn.1680935. Research Policy, Forthcoming, available at: https://ssrn.com/abstract=1680935 Hassan, M., Shaukat, S., Shakeel, M. and Imran, M. (2012), "Interrelations between organizational culture, innovation and employee performance: evidence from banking sector of Pakistan", Pakistan Journal of Social Sciences, Vol. 32 No. 2, pp. 339-355.

Helfat, C.E. and Peteraf, M.A. (2015), Managerial cognitive capabilities and the microfoundations of dynamic capabilities", Strategic Management Journal, Vol. 36 No. 6, pp. 831-850, doi: 10.1002/smj.2247.

Helfat, C.E., Finkelstein, S., Mitchell, W., Peteraf, M., Singh, H., Teece, D. and Winter, S.G. (2007), Dynamic Capabilities: Understanding Strategic Change in Organizations, Blackwell, Malden, MA.

Hsu, C.-C., Tan, K.C., Jayaram, J. and Laosirihongthong, T. (2014), "Corporate entrepreneurship, operations core competency and innovation in emerging economies", International Journal of Production Research, Vol. 52 No. 18, pp. 5467-5483, doi: 10.1080/00207543.2014.915069.

Hu, L. and Bentler, P.M. (1999), "Cutoff criteria for fit indexes in covariance structure analysis: conventional criteria versus new alternatives", Structural Equation Modeling: A Multidisciplinary Journal, Vol. 6 No. 1, pp. 1-55, doi: 10.1080/10705519909540118.

Hu, M-.L.M., Horng, J-.S. and Sun, Y-.H.C. (2009), "Hospitality teams: knowledge sharing and service innovation performance", Tourism Management, Vol. 30 No. 1, pp. 41-50, doi: 10.1016/j. tourman.2008.04.009

Huang, X., Hsieh, J.P.-A. and He, W. (2014), "Expertise dissimilarity and creativity: the contingent roles of tacit and explicit knowledge sharing", Journal of Applied Psychology, Vol. 99 No. 5, pp. 816-830, doi: 10.1037/a0036911.

Huber, G.P. (1991), "Organizational learning: the contributing processes and the literatures", Organization Science, Vol. 2 No. 1, pp. 88-115, doi: 10.1287/orsc.2.1.88. lacobucci, D. (2010), "Structural equations modeling: fit indices, sample size, and advanced topics", Journalof Consumer Psychology, Vol. 20 No. 1, pp. 90-98, doi: 10.1016/j.jcps.2009.09.003.

Jimenez-Jimenez, D., Martínez-Costa, M. and Sanz-Valle, R. (2014), "Knowledge management practices for innovation: a multinational corporation's perspective", Journal of Knowledge Management, Vol. 18 No. 5, pp. 905-918, doi: 10.1108/JKM-06-2014-0242.

Joreskog, K. and Sorbom, D. (1989), LISREL VII. Analysis of Linear Structural Relationships by Maximum Likelihood, Instrumental Variables, and Least-Squares Methods, Scientific Software, Mooresville, IN.

Joreskog, K. (2005), Structural equation modeling with ordinal variables using LISREL. Scientific Software International, Chicago, available at: www.ssicentral.com/lisrel/advancedtopics.html (accessed 29 May 2021).

Juliao-Rossi, J., Forero-Pineda, C., Losada-Otalora, M. and Peňa-García, N. (2019), "Trajectories of innovation: a new approach to studying innovation performance", Journal of Business Research, Vol. 115 No. 40, pp. 30-41, doi: 10.1016/j.jbusres.2019.11.026.

Kalko, M.M., Erena, O.T. and Debele, S.A. (2022), "Technology management practices and innovation: empirical evidence from medium- and large-scale manufacturing firms in Ethiopia", African Journal of Science, Technology, Innovation and Development, doi: 10.1080/20421338.2022.2040828.

Khan, M.N. and Zaman, U. (2020), "The effect of knowledge management practices on organizational innovation: moderating role of management support", Journal of Public Affairs, Vol. 21, No3, pp. 1-9, doi: 10.1002/pa.2284.

Koc, T., Kurt, K. and Akbiyik, A. (2019), "A brief summary of knowledge management domain: 10-Year history of the journal of knowledge management", Procedia Computer Science, Vol. 158, pp. 891-898, doi: 10.1016/j.procs.2019.09.128.

Kogut, B. and Zander, U. (1992), "Knowledge of the firm, combinative capabilities, and the replication of technology", Organization Science, Vol. 3 No. 3, pp. 383-397, doi: 10.1287/orsc.3.3.383

Kulkarni, U.R., Ravindran, S. and Freeze, R. (2007), "A knowledge management success model: theoretical development and empirical validation", Journal of Management Information Systems, Vol. 23 No. 3, pp. 309-347, doi: 10.2753/mis0742-1222230311.

Langton, N., Robbins, S.P. and Judge, T.A. (2013), Fundamentals of Organizational Behaviour, Pearson Education.

Lee, H. and Choi, B. (2003), "Knowledge management enablers, processes, and organizational performance: an integrative view and empirical examination", Journal of Management Information Systems, Vol.20No. 1,pp. 179-228, doi: 10.1080/07421222.2003.11045756.

Lee, S., Gon Kim, B. and Kim, H. (2012), "An integrated view of knowledge management for performance", Journal of Knowledge Management, Vol. 16 No. 2, pp. 183-203, doi: 10.1108/ 13673271211218807.

Lee, K.C., Lee, S. and Kang, I.W. (2005), "KMPI: measuring knowledge management performance", Information & Management, Vol. 42 No. 8, pp. 469-482, doi: 10.1016/j.im.2004.02.003.

Lei, H., Ha, A.T.L. and Le, P.B. (2020), "How ethical leadership cultivates radical and incremental innovation: the mediating role of tacit and explicit knowledge sharing", Journal of Business & Industrial Marketing, Vol. 35 No. 5, pp. 849-862, doi: 10.1108/jbim-05-2019-0180.

Le, P.B. and Lei, H. (2019), "Determinants of innovation capability: the roles of transformational leadership, knowledge sharing and perceived organizational support", Journal of Knowledge Management, Vol. 23 No. 3, pp. 527-547, doi: 10.1108/JKM-09-2018-0568.

Le, P.B., Lei, H. and Than, T.S. (2018), "How leadership and trust in leaders foster employees' behavior toward knowledge sharing", Social Behavior and Personality: An International Journal, Vol. 46 No. 5, pp. 705-720, doi: 10.2224/sbp.6711.

Lemon, M. and Sahota, P.S. (2004), "Organizational culture as a knowledge repository for increased innovative capacity", Technovation, Vol. 24 No. 6, pp. 483-498, doi: 10.1016/s0166-4972(02)00102-5.

Leonard, D. and Sensiper, S. (1998), "The role of tacit knowledge in group innovation", California Management Review, Vol. 40 No. 3, pp. 112-132, doi: 10.2307/41165946.

Leonard, D. (1995), Wellsprings of Knowledge: Building and Sustaining the Sources of Innovation, Harvard Business School Press, Boston, MA.

Li, W., Bhutto, T.A., Nasiri, A.R., Shaikh, H.A. and Samo, F.A. (2018), "Organizational innovation: the role of leadership and organizational culture", International Journal of Public Leadership, Vol. 14 No. 1, pp. 33-47, doi: 10.1108/IJPL-06-2017-0026.

Liao, S.-H. and Wu, C.-C. (2010), "System perspective of knowledge management, organizational learning, and organizational innovation", Expert Systems with Applications, Vol. 37 No. 2, pp. 1096-1103, doi: 10.1016/j.eswa.2009.06.109.

Lin, H. (2007), "Knowledge sharing and firm innovation capability: an empirical study", International Journal of Manpower, Vol. 28 Nos 3/4, pp. 315-332, doi: 10.1108/01437720710755272.

Lin, H.F. (2006), "Impact of organizational support on organizational intention to facilitate knowledge sharing", Knowledge Management Research & Practice, Vol. 4 No. 1, pp. 26-35, doi: 10.1057/palgrave. kmrp.8500083.

Lin, S.-W. and Lo, L.Y.-S. (2015), "Mechanisms to motivate knowledge sharing: integrating the reward systems and social network perspectives", Journal of Knowledge Management, Vol. 19 No. 2, pp. 212-235, doi: 10.1108/JKM-05-2014-0209.

Loehlin, J. (2004), Latent Variable Models: an Introduction to Factor, Path, and Structural Analysis, 4th ed., Lawrence Erlbaum Associates

McDermott, R. and O'Dell, C. (2001), "Overcoming cultural barriers to sharing knowledge", Journal of Knowledge Management, Vol. 5 No. 1, pp. 76-85, doi: 10.1108/13673270110384428.

MacKinnon, D.P., Coxe, S. and Baraldi, A.N. (2012), "Guidelines for the investigation of mediating variables in business research", Journal of Business and Psychology, Vol. 27 No. 1, pp. 1-14, doi: 10.1007/s10869-011-9248-z.

Mageswari, U., Sivasubramanian, C. and Dath, S. (2017), "A comprehensive analysis of knowledge management in Indian manufacturing companies", Journal of Manufacturing Technology Management, Vol. 28 No. 4, doi: 10.1108/JMTM-08-2016-0107.

Makri M. and Scandura T.A. (2010), "Exploring the effects of creative CEO leadership on innovation in high-technology firms", The Leadership Quarterly, Vol. 21 No. 1, pp. 78-88, doi: 10.1016/j. leaqua.2009.10.006.

Mageswari, S.D.U. (2014), "Knowledge management practices in manufacturing companies in Tiruvallur district, Tamil Nadu", Doctoral dissertation, Pondicherry University

Martelo-Landroguez, S. and Cegarra-Navarro, J.-G. (2014), "Linking knowledge corridors to customer value through knowledge processes", Journal of Knowledge Management, Vol. 18 No. 2, pp. 342-365, doi: 10.1108/JKM-07-2013-0284.

Masa'deh, R., Obeidat, B.Y. and Tarhini, A. (2016), "A Jordanian empirical study of the associations among transformational leadership, transactional leadership, knowledge sharing, job performance, and firm performance: a structural equation modelling approach", Journal of Management Development, Vol. 35 No. 5, pp. 681-705, doi: 10.1108/JMD-09-2015-0134.

Mokhber, M., Khairuzzaman, W. and Vakilbashi, A. (2017), "Leadership and innovation: the moderator role of organization support for innovative behaviors", Journal of Management & Organization, Vol. 24 No. 1, pp. 108-128, doi: 10.1017/jmo.2017.26.

Moorman, C. and Rust, R.T. (1999), "The role of marketing", Journal of Marketing, Vol. 63 No. 4, pp. 180-197, doi: 10.1177/00222429990634s117.

Murat Ar, I. and Baki, B. (2011), "Antecedents and performance impacts of product versus process innovation: empirical evidence from SMEs located in Turkish science and technology parks", European Journal of Innovation Management, Vol. 14 No. 2, pp. 172-206, doi: 10.1108/14601061111124885.

Naranjo-Valencia, J.C., Jimenez-Jimenez, D. and Sanz-Valle, R. (2011), "Innovation or imitation? The role of organizational culture", Management Decision, Vol. 49 No. 1, pp. 55-72, doi: 10.1108/00251741111094437.

Neely, A. and Hii, J. (1998), "Innovation and business performance: a literature review", Report Produced for Government Office for the Eastern Region, The Judge Institute of Management Studies, University of Cambridge, Cambridge.

Nonaka, I. and Takeuchi, H. (1995), The Knowledge-Creating Company: How Japanese Companies Create the Dynamics of Innovation, Oxford University Press, New York, NY.

Nonaka I. and Konno N. (1998), "The concept of 'Ba': building a foundation for knowledge creation", California Management Review, Vol. 40 No. 3, pp. 40-54, doi: 10.2307/41165942

Nonaka, I. (1994), "Adynamic theory of organizational knowledge creation", Organization Science, Vol. 5 No. 1, pp. 14-37, doi: 10.1287/orsc.5.1.14.

Nonaka, I. (2007), "The knowledge-creating company", Harvard Business Review, Vol. 85 Nos 7/8, pp. 162-171.

Nonaka, I., Hirose, A. and Takeda, Y. (2016), "Meso-foundations of dynamic capabilities: team-level synthesis and distributed leadership as the source of dynamic creativity", Global Strategy Journal, Vol. 6 No. 3, pp. 168-182, doi: 10.1002/gsj.1125.

Nonaka, I., Toyama, R. and Konno, N. (2000), "SECI, Ba and leadership: a unified model of dynamic knowledge creation", Long Range Planning, Vol. 33 No. 1, pp. 5-34, doi: 10.1016/s0024-6301(99)00115-6.

Noruzy, A., Dalfard, V.M., Azhdari, B., Nazari-Shirkouhi, S. and Rezazadeh, A. (2013), "Relations between transformational leadership, organizational learning, knowledge management, organizational innovation, and organizational performance: an empirical investigation of manufacturing firms", The International Journal of Advanced Manufacturing Technology, Vol. 64 Nos 5/8, pp. 1073-1085, doi: 10.1007/s00170-012-4038-y.

Oqubay, A. (2018). The structure and performance of the Ethiopian manufacturing sector, Working Paper Series No. 299, African Development Bank, Abidjan, Cote d'Ivoire.

Oyemomi, O., Liu, S., Neaga, I., Chen, H. and Nakpodia, F. (2018), "How cultural impact on knowledge sharing contributes to organizational performance: using the fsQCA approach", Journal of Business Research, Vol. 94, pp. 313-319, doi: 10.1016/j.jbusres.2018.02.027.

Papa, A., Dezi, L., Gregori, G.L., Mueller, J. and Miglietta, N. (2020), "Improving innovation performance through knowledge acquisition: the moderating role of employee retention and human resource management practices", Journal of Knowledge Management, Vol. 24 No. 3, pp. 589-605, doi: 10.1108/JKM-09-2017-0391.

Park, H.Y., Misra, K., Reddy, S. and Jaber, K. (2019), "Family firms' innovation drivers and performance: a dynamic capabilities approach", Journal of Family Business Management, Vol. 9 No. 1, pp. 4-23, doi: 10.1108/JFBM-11-2017-0039.

Pawlowsky, P. and Schmid, S. (2012), "Interrelations between strategic orientation, knowledge management, innovation and performance. Empirical findings from a national survey in Germany", International Journal of Knowledge Management Studies, Vol. 5 No. 1/2, pp. 185-209, doi: 10.1504/ IJKMS.2012.051911.

Pedersen, E.R.G., Gwozdz, W. and Hvass, K.K. (2018), "Exploring the relationship between business model innovation, corporate sustainability, and organizational values within the fashion industry", Journal of Business Ethics, Vol. 149 No. 2, pp. 267-284, doi: 10.1007/s10551-016-3044-7.

Prajogo, D.I. and Sohal, A.S. (2006), "The integration of TQM and technology/R&D management in determining quality and innovation performance", Omega, Vol. 34 No. 3, pp. 296-312, doi: 10.1016/j. omega.2004.11.004.

Prajogo, D.I. and McDermott, C.M. (2011), "The relationship between multidimensional organizational culture and performance", International Journal of Operations & Production Management, Vol. 31 No. 7, pp. 712-735, doi: 10.1108/01443571111144823.

Prasad, B. and Junni, P. (2016), "CEO transformational and transactional leadership and organizational innovation: the moderating role of environmental dynamism", Management Decision, Vol. 54 No. 7, pp. 1542-1568, doi: 10.1108/MD-11-2014-0651.

Rajan, R., Dhir, S and Sushil. (2021), "Technology management for innovation in organizations: an argumentation-based modified TISM approach", Benchmarking: An International Journal, Vol. 28 No. 6, pp. 1959-1986, doi: 10.1108/BIJ-01-2020-0019.

Rastogi, P.N. (2000), "Knowledge management and intellectual capital: the new virtuous reality of competitiveness", Human Systems Management, Vol. 19 No. 1, pp. 39-48, doi: 10.3233/HSM-2000-19105

Redmond, M.R., Mumford, M.D. and Teach, R. (1993), "Putting creativity to work: effects of leader behavior on subordinate creativity", Organizational Behavior and Human Decision Processes, Vol. 55 No. 1, pp. 120-151, doi: 10.1006/obhd.1993.1027.

Ribiere, V.M and Sitar, A.S. (2003), "Critical role of leadership in nurturing a knowledge-supporting culture", Knowledge Management Research & Practice, Vol. 1 No. 1, pp. 39-48, doi: 10.1057/palgrave. kmrp.8500004.

Ritala, P., Husted, K., Olander, H. and Michailova, S. (2018), "External knowledge sharing and radical innovation: the downsides of uncontrolled openness", Journal of Knowledge Management, Vol. 22 No. 5, pp. 1104-1123, doi: 10.1108/JKM-05-2017-0172.

Rosing, K., Frese, M. and Bausch, A. (2011), "Explaining the heterogeneity of the leadership-innovation relationship: ambidextrous leadership", The Leadership Quarterly, Vol. 22 No. 5, pp. 956-974, doi: 10.1016/j.leaqua.2011.07.014.

Rossi, I., Bertassini, A.C., Ferreira, C., Neves do Amaral, W.A. and Ometto, A.R. (2019), "Circular economy indicators for organizations considering sustainability and business models: plastic, textile and electro-electronic cases", Journal of Cleaner Production, Vol. 247, doi: 10.1016/j. jclepro.2019.119137.

Sackmann, S. (1992), "Culture and subcultures: an analysis of organizational knowledge", Administrative Science Quarterly, Vol. 37 No. 1, pp. 140-161, doi: 10.2307/2393536.

Schumacker, R. and Lomax, R. (2016), A Beginner's Guide to Structural Equation Modelling, 4th ed., Routledge, Taylor & Francis Group, New York, NY.

Senge, P.M. (1990), The Fifth Discipline, Doubleday, New York, NY.

Shanker, R., Bhanugopan, R., Van der Heijden, B.I. and Farrell, M. (2017), "Organizational climate for innovation and organizational performance: the mediating effect of innovative work behavior", Journal of Vocational Behavior, Vol. 100, pp. 67-77, doi: 10.1016/j.jvb.2017.02.004.

Sheehan, M., Garavan, T.N. and Morley, M.J. (2020), "Transformational leadership and work unit innovation: a dyadic two-wave investigation", Journal of Business Research, Vol. 109, pp. 399-412, doi: 10.1016/j.jbusres.2019.10.072.

Sheng, M.L. (2017), "A dynamic capabilities-based framework of organizational sensemaking through combinative capabilities towards exploratory and exploitative product innovation in turbulent environments", Industrial Marketing Management, Vol. 65 No. 2, pp. 28-38, doi: 10.1016/j. indmarman.2017.06.001.

Singh, S.K., Del Giudice, M., Tarba, S.Y. and De Bernardi, P. (2019), "Top management team shared leadership, market-oriented culture, innovation capability, and firm performance", IEEE Transactions on Engineering Management,, doi: 10.1109/TEM.2019.2946608.

Singh, S.K., Gupta, S., Busso, D. and Kamboj, S. (2019), "Top management knowledge value, knowledge sharing practices, open innovation and organizational performance", Journal of Business Research, Vol. 128, doi: 10.1016/j.jbusres.2019.04.040.

Skerlavaj, M., Song, J. and Lee, Y. (2010), "Organizational learning culture, innovative culture and innovations in South Korean firms", Expert Systems with Applications, Vol. 37 No. 9, pp. 6390-6403, doi: 10.1016/j.eswa.2010.02.080.

Spender, J.-C. and Grant, R.M. (1996), "Knowledge and the firm: overview", Strategic Management Journal, Vol. 17 No. S2, pp. 5-9, doi: 10.1002/smj.4250171103.

Steiger, J. and Lind, J. (1980), "Statistically based tests for the number of common factors", Paper presented at the annual meeting of the Psychometric Society, IA City, IA.

Tamer Cavusgil, S., Calantone, R.J. and Zhao, Y. (2003), "Tacit knowledge transfer and firm innovation capability", Journal of Business & Industrial Marketing, Vol. 18 No. 1, pp. 6-21, doi: 10.1108/08858620310458615.

Teece, D.J. (1996), "Firm organization, industrial structure, and technological innovation", Journal of Economic Behavior & Organization, Vol. 31 No. 2, pp. 193-224, doi: 10.1016/S0167-2681(96) 00895-5.

Teece, D.J. Jr. (1998), "Design issues for innovative firms: bureaucracy, incentives and industrial structure", in Chandler, A.D. Haagstrom, P. and Solvell, O. (Eds), The Dynamic Firm: The Role of Technology, Strategy, Organization, and Regions, Oxford University Press, Oxford.

Teece, D.J. (2000), "Strategies for managing knowledge assets: the role of firm structure and industrial context", Long Range Planning, Vol. 33 No. 1, pp. 35-54, doi: 10.1016/S0024-6301(99) 00117-X.

Teece, D.J. (2007), "Explicating dynamic capabilities: the nature and microfoundations of (sustainable) enterprise performance", Strategic Management Journal, Vol. 28 No. 13, pp. 1319-1350, doi: 10.1002/ smj.640.

Teece, D.J. (2010), "Business models, business strategy and innovation", Long Range Planning, Vol. 43 No. 2, pp. 172-194, doi: 10.1016/j.lrp.2009.07.003.

Teece, D.J., Pisano, G. and Shuen, A. (1997), "Dynamic capabilities and strategic management", Strategic Management Journal, Vol. 18 No. 7, pp. 509-533, doi: 10.1002/(SICI)1097-0266(199708) 18:7<509::AID-SMJ882>3.0.CO;2-Z

Tellis, G.J. (2012), Unrelenting Innovation: How to Create a Culture for Market Dominance, John Wiley and Sons, San Francisco, CA.

Tidd, J., Bessant, J. and Pavitt, K. (2005), Managing Innovation: Integrating Technological, Market and Organizational Change, 3rd ed., Wiley, Chichester.

UNCTAD (2015), "Technology and innovation report", United nations publication unctad/tir/2015 ISSN 2076-2917.

UNDP (2018), Understanding African Experiences in Formulating and Implementing Plans for Emergence: A Case Study of Growing Manufacturing Industry in Ethiopia, UNDP, Addis Ababa.

Valencia, J.C.N., Valle, R.S. and Jimenez, D.J. (2010), "Organizational culture as a determinant of product innovation", European Journal of Innovation Management, Vol. 13 No. 4, pp. 466-480, doi: 10.1108/14601061011086294.

Vargas, M.I.R. (2015), "Determinant factors for small business to achieve innovation, high performance and competitiveness: organizational learning and leadership style", Procedia - Social and Behavioral Sciences, Vol. 169, pp. 43-52, doi: 10.1016/j.sbspro.2015.01.284.

Vicente, G.-N., Oltra-Badenesa, R., Gil-Gomez, H. and Fernández, A.I. (2021), "Customer relationship management (CRM) and innovation: a qualitative comparative analysis (QCA) in the search for improvements on the firm performance in winery sector", Technological Forecasting & Social Change, Vol. 169, doi: 10.1016/j.techfore.2021.120838.

Wakeford, J.J., Gebreeyesus, M., Ginbo, T., Yimer, K., Manzambi, O., Okereke, C., Black M. and Mulugetta, Y. (2017), "Innovation for green industrialisation: an empirical assessment of innovation in Ethiopia's cement, leather and textile sectors", Journal of Cleaner Production, Vol. 166, pp. 503-511, doi: 10.1016/j.jclepro.2017.08.067.

Wan, D., Ong, C.H. and Lee, F. (2005), "Determinants of firm innovation in Singapore", Technovation, Vol. 25 No. 3, pp. 261-268, doi: 10.1016/s0166-4972(03)00096-8.

Wang, S. and Noe, R.A. (2010), "Knowledge sharing: a review and directions for future research", Human Resource Management Review, Vol. 20 No. 2, pp. 115-131, doi: 10.1016/j.hrmr.2009.10.001.

Wang, D., Su, Z. and Yang, D. (2011), "Organizational culture and knowledge creation capability", Journal of Knowledge Management, Vol. 15 No. 3, pp. 363-373, doi: 10.1108/13673271111137385.

Wang, Z., Wang, N. and Liang, H. (2014), "Knowledge sharing, intellectual capital and firm performance", Management Decision, Vol. 52 No. 2, pp. 230-238, doi: 10.1108/MD-02-2013-0064.

Wheaton, B., Muthen, B., Alwin, D. and Summers, G. (1977), "Assessing reliability and stability in panel models", Sociological Methodology, Vol. 8, pp. 84-136, doi: 10.2307/270754.

Wolf, E.J., Harrington, K.M., Clark, S.L. and Miller, M.W. (2013), "Sample size requirements for structural equation models: an evaluation of power, bias, and solution propriety", Educational and Psychological Measurement, Vol. 73 No. 6, pp. 1-20, doi: 10.1177/0013164413495237.

Woodfield, P. and Husted, K. (2017), "Intergenerational knowledge sharing in family firms: case-based evidence from the New Zealand wine industry", Journal of Family Business Strategy, Vol. 8 No. 1, pp. 57-69, doi: 10.1016/j.jfbs.2017.01.001.

Xiao, Y., Zhang, X. and Ordánez de Pablos, P. (2017), "How does individuals' exchange orientation moderate the relationship between transformational leadership and knowledge sharing?", Journal of Knowledge Management, Vol. 21 No. 6, pp. 1622-1639, doi: 10.1108/JKM-03-2017-0120.

Yang, Z., Nguyen, V.T. and Le, P.B. (2018), "Knowledge sharing serves as a mediator between collaborative culture and innovation capability: empirical research", Journal of Business & Industrial Marketing, Vol. 33 No. 7, pp. 958-969, doi: 10.1108/JBIM-10-2017-0245.

Ye§il, S., Buyukbese, T. and Koska, A. (2013), "Exploring the link between knowledge sharing enablers, innovation capability and innovation performance", International Journal of Innovation Management, Vol. 17 No. 4, pp. 1-20, doi: 10.1142/S1363919613500187.

Zack, M., McKeen, J. and Singh, S. (2009), "Knowledge management and organizational performance: an exploratory analysis", Journal of Knowledge Management, Vol. 13 No. 6, pp. 392-409, doi: 10.1108/ 13673270910997088.

Zhang, X. and Jiang, J.Y. (2015), "With whom shall I share my knowledge? A recipient perspective of knowledge sharing", Journal of Knowledge Management, Vol. 19 No. 2, pp. 277-295, doi: 10.1108/JKM-05-2014-0184.

Zheng, J., Wu, G. and Xie, H. (2017), "Impacts of leadership on project-based organizational innovation performance: the mediator of knowledge sharing and moderator of social Capital", Sustainability, Vol. 9 No. 10, pp. 1-22, doi: 10.3390/su9101893.

Zheng, W., Yang, B. and McLean, G.N. (2010), "Linking organizational culture, structure, strategy, and organizational effectiveness: mediating role of knowledge management", Journal of Business Research, Vol. 63 No. 7, pp. 763-771, doi: 10.1016/j.jbusres.2009.06.005.

Zhou, K. and Wu, F. (2010), "Technological capability, strategic flexibility, and product innovation", Strategic Management Journal, Vol. 31 No. 5, pp. 547-561, doi: 10.1002/smj.830.

## Further reading

Abbas, J. and Sagsan, M. (2019), "Impact of knowledge management practices on green innovation and corporate sustainable development: a structural analysis", Journal of Cleaner Production, Vol. 229 No. 20, pp. 611-620, doi: 10.1016/j.jclepro.2019.05.024

African Union (2019), "Five-Year science, technology and innovation plan of action 2019-2024", available at: www.au.int

Argyris, C. and Schon, D.A. (1978), Organizational Learning: A Theory of Action Perspective, Addison-Wesley Publishing Company, Reading.

Argyris, C. and Schon, D.A. (1974), Theory in Practice: Increasing Professional Effectiveness, Jossey-Bass, San Francisco.

Bodell, L. and Earle, C. (2004), "The yin and yang of marketing measurement: four principles of innovation", Interactive Marketing, Vol. 6 No. 2, pp. 130-140, doi: 10.1057/palgrave.im.4340277.

Cresswell, J.W. (2009), Research Design: Qualitative, Quantitative, and Mixed Approaches, Sage Publications, Inc., Los Angeles.

European Commission (2014), "State of the innovation union", Taking Stock 2010-2014, Publications Office of the European Union, Luxembourg.

Gopalakrishnan, S. and Damanpour, F. (1997), "A review of innovation research in economics, sociology and technology management", Omega, International Journal of Management Science, Vol. 25 No. 1, pp. 15-28, doi: 10.1016/s0305-0483(96)00043-6.

Gunday, G., Ulusoy, G., Kilic, K. and Alpkan, L. (2011), "Effects of innovation types on firm performance", International Journal of Production Economics, Vol. 133 No. 2, pp. 662-676, doi: 10.1016/j.ijpe.2011.05.014.

Gupta, S., Malhotra, N., Czinkota, M. and Foroudi, P. (2016), "Marketing innovation: a consequence of competitiveness", Journal of Business Research, Vol. 69 No. 12, pp. 5671-5681, doi: 10.1016/j. jbusres.2016.02.042.

Jyoti, J., Gupta, P. and Kotwal, S. (2011), "Impact of knowledge management practices on innovative capacity", Vision: The Journal of Business Perspective, Vol. 15 No. 4, pp. 315-330, doi: 10.1177/097226291101500402.

Keskin, H. (2006), "Market orientation, learning orientation, and innovation capabilities in SMEs: an extended model", European Journal of Innovation Management, Vol. 9 No. 4, pp. 396-417, doi: 10.1108/14601060610707849.

Lin, T., Wu, S. and Lu, C. (2012), "Exploring the affect factors of knowledge sharing behavior: the relations model theory perspective", Expert Systems with Applications, Vol. 39 No. 1, pp. 751-764, doi: 10.1016/j. eswa.2011.07.068.

Lu, C.-S., Lai, K.-H. and Cheng, T.C.E. (2007), "Application of structural equation modeling to evaluate the intention of shippers to use internet services in liner shipping", European Journal of Operational Research, Vol. 180 No. 2, pp. 845-867, doi: 10.1016/j.ejor.2006.05.001.

Malhotra, N., Nunan, D. and Birks, D. (2017), Marketing Research: An Applied Approach, 5th ed., Prentice Hall Inc.

Naqshbandi, M.M. and Jasimuddin, S.M. (2018), "Knowledge-oriented leadership and open innovation: role of knowledge management capability in France-based multinationals", International Business Review, Vol. 27 No. 3, pp. 701-713, doi: 10.1016/j.ibusrev.2017.12.001.

Polanyi, M. (1966), "The logic of tacit inference", Philosophy, Vol. 41 No. 155, pp. 1-18, doi: 10.2307/ 3749034.

Razak, N.A., Pangil, F., Zin, M.L., Yunus, N.A. and Asnawi, N.H. (2016), "Theories of knowledge sharing behavior in business strategy", Procedia Economics and Finance, Vol. 37, pp. 545-553, doi: 10.1016/ S2212-5671(16)30163-0.

Saunila, M. and Ukko, J. (2012), "A conceptual framework for the measurement of innovation capability and its effects", Baltic Journal of Management, Vol. 7 No. 4, pp. 355-375, doi: 10.1108/ 17465261211272139.

Schein, E.H. (1990), "Organizational culture", American Psychologist, Vol. 45 No. 2, pp. 109-119, doi: 10.1037/0003-066X.45.2.109.

Schein, E.H. (2000), "Sense and nonsense about culture and climate", in Ashkanasy, N.M., Widerom, C.P.M. and Peterson M.F. (Eds), Handbook of Organizational Culture and Climate, Sage Publications, Thousand Oaks, CA, pp. 23-30.

Tether, B.S. (2003), "The sources and aims of innovation in services: variety between and within sectors", Economics of Innovation and New Technology, Vol. 12 No. 6, pp. 481-505, doi: 10.1080/ 1043859022000029221.

UNESCO (2015), Science Report, available at: https://en.unesco.org/unescosciencereport

Zhang, Z., Song, F. and Song, Z. (2019), "Promoting knowledge sharing in the workplace: punishment v. reward", Chaos, Solitons and Fractals, Vol. 30 No. 10, doi: 10.1016/j.chaos.2019.109518.



# Appendix 1

Figure A1 An organizational factors measurement model



Figure A2 KM measurement model

Appendix 2



Figure A3 Innovation measurement model



#### Figure A4 Second-order KM model



Figure A5 Second-order innovation model

### Table A1 Items to measure organizational factors, KM and innovation

		Standard loading
OGCUL		
OGCUL1	Our organization established procedures for governing daily activities	0.62
OGCUL2	The speed of knowledge exchange among our organization's employees is good	0.71
OGCUL3	In our organization, open sharing of knowledge is encouraged	0.90
OGCUL4	Our organization values openness, collaboration and trust for invention and innovation	0.76
OGCUL5	Our organization values learning, sharing and applying knowledge	0.74
LMS		
LMS1	Our top management permits and creates an atmosphere for KSHR	0.61
LMS2	Our top management emphasizes the security of jobs and stability in relationships	0.76
LMS3	Leaders/managers generate high levels of motivation and commitment in the workforce	0.91
LMS4	Our organization encourages teamwork and participation	0.67
KAC dimension		
KAC1	We attend courses, seminars or other training for skill development	0.78
KAC2	We actively participate in an outside professional network like industry associations, conferences, etc.	0.90
KAC3	Exit interviews are carried out to capture critical knowledge and experience when our employees leave our organization	0.82
KAC4	We hire consultants when important skills/expertise or information about any activity is not available in our organization	0.61
KSHR dimension		
KSHR1	Problems related to processes are discussed openly in our organization	0.67
KSHR2	Mechanisms are in place to encourage members of an organization to share information	0.77
KSHR3	There is informal communication for KSHR in our organization	0.76
KSHR4	There is a structured induction program for new employees	0.80
KCOV dimension		
KCOV1		0.75
		(continued)
		(

Table A1

Stand	lard I	loadi	ing
-------	--------	-------	-----

	We have a team to study and communicate the market scenario to the management for further action	
KCOV2	Brainstorming sessions among employees, managers and top management are frequently used for solving problems	0.80
KCOV3	We research to explore future possibilities of expansion in terms of capacity, markets, etc.	0.69
KSTR dimension	1	
KST1	Our company regularly stores knowledge (has archives) on the implementation and contents of the research process	0.74
KST2	Our company has well-organized documentation on the knowledge and achievements of employees	0.85
KST3	Following the achievement of significant work results, Our company conducts interviews with operators on the work process	0.73
PDI PRDIN1 PRDIN2 PRDIN3 PRDIN4	The level of the newness (novelty) of our firm's new products The speed of our new product development The number of new products our firm has introduced to the market The number of our new products that are first to market (early market entrants)	0.67 0.93 0.72 0.72
PCI PRCIN1 PRCIN2	The technological competitiveness of our company is high The speed at which we adopt the latest technological innovations in our processes is very	0.80 0.87
PRCIN3 PRCIN4	There is the novelty of the technology used in our processes Our processes, techniques and technology are changing at a rapid pace	0.90 0.61
ADMIN ADMIN1	We designed the ISO9001 or ISO/TS16949 QMS around its business processes	0.67
ADMIN2	We have clear, standardized and documented process instructions that are well understood by our employees	0.93
ADMIN3	variation	0.79
ADMIN4	We are engaged in an active competitive benchmarking program to measure our performance against the "best practice" in the industry	0.75
MRIN		
MRIN1	Our company is creative in its methods of marketing operations	0.72
MRIN2	Our company is frequently the first to introduce new products and services to the market	0.77
MRIN3	Our new product introduction has increased over the past five years	0.76
MRIN4 MRIN5	Our company trequently tries out new ideas Our company seeks out new ways to do things	0.77 0.67

# Appendix 3

Item	Mean	SD	Skewness	Kurtosis
OGCUL1	2.7386	0.99849	-0.137	-0.459
OGCUL2	2.8824	0.93851	-0.198	-0.049
OGCUL3	2.9804	0.89935	0.094	0.115
OGCUL4	3.0915	0.91306	0.132	-0.263
OGCUL5	3.085	0.97298	-0.129	-0.58
LMS1	3.098	1.08072	-0.324	-0.527
LMS2	3	1.02598	-0.111	-0.367
LMS3	2.7778	1.04013	0.067	-0.529
LMS4	2.9216	0.94958	-0.029	-0.335
KAC1	2.8824	0.92439	-0.269	-0.49
KAC2	2.8824	0.99301	-0.047	-0.327
KAC3	3.0784	0.89978	-0.211	0.142
KAC4	3.1569	0.98754	-0.321	-0.188
KCOV1	2.7059	0.90235	0.019	-0.19
KCOV2	2.817	0.983	0.039	-0.55
KCOV3	2.8889	1.00365	0.028	-0.541
KSHR1	3.1242	0.96191	-0.207	-0.281
KSHR2	3.0915	0.94144	-0.28	-0.359
KSHR3	2.9477	0.98536	0.022	-0.256
KSHR4	3.0327	0.94185	-0.066	-0.214
KST1	2.8627	0.93231	-0.314	-0.424
KST2	2.8562	0.98954	0.005	-0.477
KST3	2.902	0.95121	-0.22	-0.352
PRDIN1	2.7908	0.86339	-0.016	-0.046
PRDIN2	2.8235	0.98085	0.15	-0.35
PRDIN3	2.8758	0.97549	0.037	-0.308
PRDIN4	2.8889	0.94281	0.034	-0.697
PRCIN1	2.8758	0.98889	0.294	-0.123
PRCIN2	2.8824	1.08179	0.174	-0.718
PRCIN3	2.9346	1.08624	-0.025	-0.611
PRCIN4	3.0065	0.98339	0.071	-0.23
ADMIN1	2.7908	1.03016	0.284	-0.311
ADMIN2	2.8301	1.06246	0.079	-0.676
ADMIN3	2.9542	1.06586	0.059	-0.553
ADNIN4	3.0131	1.01946	-0.14	-0.456
MRIN1	3.0065	1.10321	-0.073	-0.54
MRIN2	2.9739	1.03839	0.017	-0.614
MRIN3	3.0327	1.10274	0.054	-0.439
MRIN4	3.0392	1.0443	-0.044	-0.381
MRIN5	3.0588	1.04652	-0.119	-0.456

 Table A2 Descriptive Statistics and Univariate Normality Test

# Appendix 4

(PE=Pearson Product Moment, PC=Polychoric, PS=Polyserial)						
Variable ve Variable	Corrolation	Chi Sau	DE	P Value	PIL	P.Value
variable vs variable	Correlation	Chi-Squ.	D.F.	r-value	HIVISEA	r-value
KAC2 vs LMS3	0.187 (PC)	29.745	15	0.013	0.080	0.760
KAC2 vs LMS4	0.181 (PC)	27.660	15	0.024	0.074	0.823
KAC2 vs KAC1	0.812 (PC)	17 413	15	0.295	0.032	0.986
KAC3 vs OGCUL 1	0.114 (PC)	8 737	15	0.891	0.000	1,000
KAC3 vs OGCUL2	0.193 (PC)	16 189	15	0.370	0.023	0.991
KAC3 vs OGCUL3	0.093 (PC)	19 168	15	0.206	0.043	0.974
KAC3 vs OGCULA	0.127 (PC)	27 379	15	0.026	0.073	0.831
KAC3 vs OGCUL5	0.037 (PC)	13 149	15	0.591	0.000	0.998
KAC3 vs LMS1	0.333 (PC)	8 658	15	0.895	0.000	1 000
KAC3 vs LMS2	0.278 (PC)	21 759	15	0.114	0.054	0.947
KAC3 vs LMS3	0.174 (PC)	19 179	15	0.206	0.043	0.974
KAC3 vs LMS4	0.166 (PC)	22.976	15	0.085	0.059	0.928
KAC3 vs KAC1	0.687 (PC)	24 559	15	0.056	0.065	0.899
KAC3 vs KAC2	0.816 (PC)	6.033	15	0.979	0.000	1 000
KAC4 vs OGCUL 1	0.144 (PC)	11 585	15	0.710	0.000	0.999
KAC4 vs OGCUL2	0.274 (PC)	15 706	15	0.402	0.018	0.993
KAC4 vs OGCUL3	0.098 (PC)	10.805	15	0.766	0.000	0.000
KACA vs OGCULA	0.035 (PC)	15 101	15	0.444	0.007	0.994
KACA vs OGCUL5	0.059 (PC)	22 306	15	0.100	0.056	0.039
KAC4 vs LMS1	0.320 (PC)	27.046	15	0.028	0.072	0.840
KAC4 vs LMS2	0.155 (PC)	19.073	15	0.210	0.042	0.975
KAC4 vs LMS3	0.122 (PC)	32,729	15	0.005	0.088	0.658
KAC4 vs LMS4	0.191 (PC)	23 263	15	0.079	0.060	0.923
KAC4 vs KAC1	0.430 (PC)	33 291	15	0.004	0.089	0.638
KAC4 vs KAC2	0.551 (PC)	28.592	15	0.018	0.077	0.796
KAC4 vs KAC3	0.661 (PC)	21.533	15	0.121	0.053	0.950
KCOV1 vs OGCUL1	0.312 (PC)	9.525	15	0.849	0.000	1.000
KCOV1 vs OGCUL2	0.310 (PC)	17,791	15	0.274	0.035	0.984
KCOV1 vs OGCUL3	0.110 (PC)	21,414	15	0.124	0.053	0.951
KCOV1 vs OGCUL4	0.221 (PC)	23,104	15	0.082	0.059	0.926
KCOV1 vs OGCUL5	0.322 (PC)	18.593	15	0.233	0.040	0.979
KCOV1 vs LMS1	0.344 (PC)	26.113	15	0.037	0.070	0.864
KCOV1 vs LMS2	0.246 (PC)	13.189	15	0.588	0.000	0.998
KCOV1 vs LMS3	0.385 (PC)	19.448	15	0.194	0.044	0.972
KCOV1 vs LMS4	0.355 (PC)	21.705	15	0.116	0.054	0.947
KCOV1 vs KAC1	0.314 (PC)	11.815	15	0.693	0.000	0.999
KCOV1 vs KAC2	0.339 (PC)	24.255	15	0.061	0.064	0.905
KCOV1 vs KAC3	0.347 (PC)	19.192	15	0.205	0.043	0.974
KCOV1 vs KAC4	0.359 (PC)	30.750	15	0.009	0.083	0.727
KCOV2 vs OGCUL1	0.288 (PC)	10.595	15	0.781	0.000	1.000
KCOV2 vs OGCUL2	0.255 (PC)	18.004	15	0.262	0.036	0.982
KCOV2 vs OGCUL3	0.183 (PC)	23.448	15	0.075	0.061	0.920
KCOV2 vs OGCUL4	0.214 (PC)	13.427	15	0.569	0.000	0.997
KCOV2 vs OGCUL5	0.210 (PC)	15.772	15	0.397	0.018	0.992
KCOV2 vs LMS1	0.372 (PC)	20.532	15	0.152	0.049	0.961
KCOV2 vs LMS2	0.371 (PC)	12.740	15	0.622	0.000	0.998
KCOV2 vs LMS3	0.415 (PC)	18.553	15	0.235	0.039	0.979
KCOV2 vs LMS4	0.459 (PC)	9.463	15	0.852	0.000	1.000
KCOV2 vs KAC1	0.349 (PC)	15.343	15	0.427	0.012	0.994
KCOV2 vs KAC2	0.404 (PC)	26.896	15	0.030	0.072	0.844
KCOV2 vs KAC3	0.317 (PC)	11.982	15	0.680	0.000	0.999
KCOV2 vs KAC4	0.459 (PC)	26.366	15	0.034	0.070	0.858
KCOV2 vs KCOV1	0.652 (PC)	28.469	15	0.019	0.077	0.800
KCOV3 vs OGCUL1	0.115 (PC)	15.793	15	0.396	0.019	0.992
						(continued)

## Table A3

(PE=	Corre Pearson Product=	lations and Te Moment, PC=	st Statistic =Polychor	cs ic, PS=Polyse	erial)	
Test of Model			Test of Close Fit			
Variable vs Variable	Correlation	Chi-Squ.	D.F.	P-Value	RMSEA	P-Value
KCOV3 vs OGCUL2	0.197 (PC)	14.774	15	0.468	0.000	0.995
KCOV3 vs OGCUL3	0.136 (PC)	22.810	15	0.088	0.058	0.931
KCOV3 vs OGCUL4	0.225 (PC)	17.763	15	0.275	0.035	0.984
KCOV3 vs OGCUL5	0.162 (PC)	14.118	15	0.517	0.000	0.996
KCOV3 vs LMS1	0.151 (PC)	14.043	15	0.522	0.000	0.997
KCOV3 vs LMS2	0.193 (PC)	26.660	15	0.032	0.071	0.850
KCOV3 vs LMS3	0.254 (PC)	28.829	15	0.017	0.078	0.789

Table A3

Indices statistics	Model 1	Model 2	Model 3	Recommended cut-off rule	Reference	
χ <sup>2</sup>	41.87	133.59	319.56	_		
Df	21	69	110	-		
p-value	0.004	0.000	0.000	> 0.05		
$\chi^2/df$	1.99	1.93	2.90	< 5	Wheaton et al. (1977)	
RMSEA	0.08	0.078	0.068	< 0.1	Steiger and Lind (1980)	
SRMR	-	0.062	0.045	< 0.08	Hu and Bentler (1999)	
NFI	0.96	0.91	0.90	> 0.90	Bentler and Bonett (1980)	
NNEL	-	0.94	0.90	> 0.90	Bentler and Bonett (1980)	
CFI	0.98	0.95	0.93	> 0.90	Bentler (1990)	
IFI	0.98	0.95	0.93	> 0.90	Bentler (1990)	
GFI	0.94	0.90	0.90	≥ 0.90	Jöreskog and Sörbom (1989)	
Source: Adopted from Kalko et al. (2022)						